

OVER-SIGNALING VS UNDERPRICING: THE ROLE OF FINANCIAL INTERMEDIARIES IN INITIAL PUBLIC OFFERINGS

Fabrizio Adriani

Luca G. Deidda

Silvia Sonderegger

WORKING PAPERS



2007/14

CENTRO RICERCHE ECONOMICHE NORD SUD (CRENOS) UNIVERSITÀ DI CAGLIARI UNIVERSITÀ DI SASSARI

Il CRENoS è un centro di ricerca istituito nel 1993 che fa capo alle Università di Cagliari e Sassari ed è attualmente diretto da Raffaele Paci. Il CRENoS si propone di contribuire a migliorare le conoscenze sul divario economico tra aree integrate e di fornire utili indicazioni di intervento. Particolare attenzione è dedicata al ruolo svolto dalle istituzioni, dal progresso tecnologico e dalla diffusione dell'innovazione nel processo di convergenza o divergenza tra aree economiche. Il CRENoS si propone inoltre di studiare la compatibilità fra tali processi e la salvaguardia delle risorse ambientali, sia globali sia locali.

Per svolgere la sua attività di ricerca, il CRENOS collabora con centri di ricerca e università nazionali ed internazionali; è attivo nell'organizzare conferenze ad alto contenuto scientifico, seminari e altre attività di natura formativa; tiene aggiornate una serie di banche dati e ha una sua collana di pubblicazioni.

www.crenos.it info@crenos.it

> CRENOS - CAGLIARI VIA SAN GIORGIO 12, I-09100 CAGLIARI, ITALIA TEL. +39-070-6756406; FAX +39-070- 6756402

> CRENOS - SASSARI VIA TORRE TONDA 34, I-07100 SASSARI, ITALIA TEL. +39-079-2017301; FAX +39-079-2017312

Titolo: OVER-SIGNALING VS UNDERPRICING: THE ROLE OF FINANCIAL INTERMEDIARIES IN INITIAL PUBLIC OFFERINGS?

ISBN: 978-88-8467-435-7

Prima Edizione: Dicembre 2007

© CUEC 2007 Via Is Mirrionis, 1 09123 Cagliari Tel./Fax 070291201 www.cuec.it

Over-signaling vs Underpricing: The Role of Financial Intermediaries in Initial Public Offerings*

Fabrizio Adriani CeFiMS, SOAS, University of London

Luca G. Deidda CeFiMS, SOAS, University of London, CRENoS, and Università di Sassari

> Silvia Sonderegger University of Bristol

This version: December 2007

Abstract

We consider a model of Initial Public Offerings (IPOs) where issuing firms of better quality are more reluctant to go public. IPOs either generate or destroy value depending on the type of the issuing firm, which is only observed by the issuer. We show that, when the issuer directly offers the shares to the investors, market breakdown occurs. This is caused by the issuer's attempts to signal his type through the offering price. Things change if we introduce a financial intermediary which: 1) acts as an underwriter, 2) influences the offering price. Underwriting creates a wedge between the interests of the intermediary and those of the issuer. This allows trade with outside investors to be restored. A by-product of the conflict of interest between issuer and intermediary is that trade is characterized by underpricing. In the benchmark case where her profits are zero, the intermediary acts as a screening device: she underwrites the shares only upon receiving positive information about the issuer.

JEL CODES: G24, G32

KEYWORDS: IPO, Signaling, Financial Intermediary, Underwriting, Underpricing.

^{*}We thank Francesco Giovannoni, Paul Grout, Bill Megginson, In-Uck Park, Lihui Tian, and seminar participants at CeFiMS, and Bristol-CMPO for comments. Corresponding author: Fabrizio Adriani, SOAS, University of London, Thornhaugh Street, London, WC1H 0XG, United Kingdom. E-Mail: fa30@soas.ac.uk

1 Introduction

In most countries, initial public offerings (IPOs) are characterized by the presence of a financial intermediary. Typically, the intermediary acts as the underwriter and plays an active role in setting the offering price at which the shares are to be issued to the public. If the issuing firm and the intermediary find an agreement so that the IPO takes place, the firm receives the proceeds of the issue net of the commission fee, or gross spread, to be paid to the intermediary. Researchers have paid little attention to the analysis of this intermediation process. In particular, the existing literature has largely ignored the question of why issuing firms do not seem able to do without intermediaries.

This paper aims at offering a theoretical justification of why IPOs are characterized by the presence of intermediaries. Since intermediaries are a pervasive feature of IPOs, a better understanding on their role and incentives is clearly important. We develop a model where, consistent with what happens in practice, the intermediary (1) underwrites shares and (2) sets the offering price in exchange for a fee conditional on the IPOs taking place. We show that, in the absence of an intermediary performing these functions, the market for IPOs may not be viable. By converse, the presence of an intermediary performing these two functions can ensure a positive amount of trade. This occurs because the intermediary's dual role as a buyer and a seller creates a wedge between her interests and those of the issuer. A by-product of this conflict of interest is that trade is characterized by underpricing.

More precisely, we consider a model of fixed price offers. There are two types of issuers: those who own good (high quality) firms and those who own bad (low quality) firms. The issuer knows his type, while investors observe a private noisy signal. Owners of good firms are more reluctant to go public than owners of bad firms since their outside option – namely, keeping full ownership of the firm – is more valuable. Moreover, going public generates gains from trade only if the firm is good. When firms are bad, trade is socially inefficient.

Under these assumptions, we compare IPOs in which the issuer directly offers shares to the investors with situations in which the offering is managed by a financial intermediary.

Since the issuer has more information than the investors, an adverse selection problem emerges. Bad firms may want to mimic good firms. We show that this adverse selection problem is exacerbated by the issuer's attempt to signal his type through the choice of the offering price. Signaling creates an upward pressure on the offering price that eventually causes market breakdown. The intuition is that good types of issuers tend to raise the offering price to differentiate themselves. Bad types raise the offering price to mimic. This signaling spiral only stops when the offering price is too high for trade to occur. The behavior of the issuer is thus characterized by what we call "over-signaling": the issuer would be better off by committing to offering prices that do not depend on his information. In this case, a positive amount of trade would be possible.

Since issuing securities directly to the investors is not a viable option for the issuer, we ask whether a financial intermediary (investment bank) can do better. The investment bank

¹As Ritter 2003, page 255, puts it "When a firm decides to issue securities to the public, it almost always hires an intermediary, typically an investment banking firm."

underwrites the shares and is able to influence the offering price. We identify a key trade-off.

On the one hand, the use of an intermediary may ensure that a positive amount of trade is restored. Since the investment bank underwrites the shares, a conflict of interest emerges between the bank and the issuer. While all the benefits from a high offering price accrue to the issuer, the investment bank bears the cost of subscribing potentially overpriced shares. The presence of this conflict between the bank and the issuer reduces the upward pressure on the offering price, preventing it from spiralling as in the case of a direct issue. As a result, there is no over-signaling and trade is therefore possible. This happens even when the investment bank is privately informed.

On the other hand, we show that the intermediary always makes losses from her underwriting activities. Intuitively, the intermediary suffers from a "seller's curse". The shares that she manages to sell on behalf of the issuer are, on average, underpriced. In contrast, the shares that she doesn't manage to sell – and for which she has to pay herself – are, on average, overpriced. As a result, the intermediary is only viable if the underwriting fee she obtains from the issuer is sufficiently high. This implies that, although the presence of the investment bank may allow trade to occur, part of the gains from trade must be used to finance her loss-making underwriting activities.

The question then arises, whether it is possible to simultaneously restore trade, and keep the investment bank from making losses. We show that this is indeed the case, by fully characterizing the natural benchmark in which the investment bank makes exactly zero profits. In this equilibrium, trade between the issuer and the investor is possible. Moreover, underpricing is particularly severe since the offering price is the lowest compatible with participation by the issuer. Another characteristic of the equilibrium is that the underwriter acts as a screening device, weeding out those firms that are most likely to be of low quality. The bank only underwrites shares over which she receives favorable information. This function of the investment bank endogenously emerges in our model even in the absence of the reputation concerns or other motives that could apply in a multi-period setting.

The paper also discusses the implications of the presence of the intermediary for the abnormal first day returns documented by the literature (underpricing). We identify two channels through which our model generates systematic underpricing. First, the seller's curse suffered by the intermediary vis-à-vis the investor implies that, conditional on the investor purchasing the shares, these are on average priced below their value. Second, we show that shares that are traded are on average underpriced even when the intermediary's informational disadvantage with respect to the investor – and therefore the seller's curse – disappears. This is a direct result of the intermediary's dual role as a buyer and a seller. Independently of the information structure, the very presence of an underwriter has a negative impact upon the offering price. When the intermediary suffers from an informational disadvantage, this mitigates the effect of the seller's curse on the offering price. In standard models of adverse selection the seller's curse would make sellers more reluctant to trade, thus driving prices upwards. In our setting this logic applies only in part. On the one hand, a more pronounced seller's curse increases the underwriting fee that the intermediary requires to break even. This in turn exerts an upward pressure on the price at which the issuer, who pays the fee,

is willing to participate. On the other hand, the intermediary is also a buyer and therefore suffers from paying an excessively high price to the issuer. This exerts a downward pressure on the offering price.

Finally, we show that the extent of underpricing is positively correlated with the investors' decision to subscribe the shares. This matches the evidence of Cornelli and Goldreich (2003) who document a positive correlation between first day return and oversubscription.

The paper is organized as follows. The next section briefly reviews the existing literature. Section three informally discusses the key features of the environment. Section four outlines the model. Section five presents the result that direct issues fail. Section six focuses on intermediated issues. Section seven discusses the model's implications. Section eight addresses possible extensions and robustness. A final section presents conclusions and discusses future research. All proofs can be found in the Appendix.

2 Related literature

The literature on IPOs consists of various strands, including the relationship between issuers and intermediaries, the reasons to go public and, of course, underpricing. The present paper puts forward a teory that justifies intermediation services in IPOs – namely underwriting and pricing. Baron (1982) analyses the optimal delegation contract between an issuer and an investment bank. The issuer faces problems of both adverse selection and moral hazard. The investment bank has private information about demand (and therefore potential proceeds from the issue), and her distribution effort is unobservable by the issuer. Under the optimal contract, the intermediary operates both an advisory role (about the price) and a distribution role. Hölmstrom and Baron (1980) study the optimal contract in the context of negotiated sales where the asymmetry of information in favour of the banker emerges only after the contracting stage due to pre-selling activities. More recently, Biais, Bossaerts, and Rochet (2002) have analyzed the IPO mechanism that maximizes proceeds from sales in a setting where the issuer is again the less informed party.

In these contributions, the intermediary's superior information provides a natural justification for intermediation. By contrast, we focus on uncertainty about the quality of the prospects of issuing firms. This is known by the issuer but only imperfectly observed by the intermediary (and by investors). Hence, in our setting, the case for an intermediary is less obvious, since the intermediary does not possess superior information. Chemmanur and Fulghieri (1994) also focus on problems arising from uncertainty about the quality of the issuing firm. In their model, investment banks evaluate entrepreneurs' projects and report to investors in return for a fee. This costly activity is prone to moral hazard. Accordingly, investment banks' credibility depends on their history. Reputation concerns provide investment banks with the incentives to collect information and reveal it truthfully. A similar point is made by Sherman (1999) who allows for both reputation and litigation costs. Our

²Baron (1979) studies pricing and distribution of issuers when banks' distribution effort is unobservable and agents are risk averse. In that context, the optimal contract is such that issuer sacrifices some of the gains from risk sharing in order to provide the banker with the right incentives to distribute the issue.

results can be seen as complementary to theirs, since we show that underwriting can play a similar role to reputation or litigation costs in aligning the intermediary's incentives with those of the investors.

Our paper is also related to the underpricing literature. According to Ritter and Welch (2002, p.11), there are probably "[..] no exceptions to the rule that the IPOs of operating companies are underpriced, on average, in all countries [..]". The existing literature on underpricing in IPOs is extensive and has offered various explanations for such stylized fact. Under the optimal contract studied by Baron (1982), the issuer has to sacrifice part of the proceeds from the issue to provide the investment bank with the right incentives. However, this result is questioned by Muscarella and Vetsuypens (1989) on empirical grounds. They find that even in the case of investment banks going public – a situation in which asymmetric information about the demand for the issue should not be relevant – IPOs are still characterized by significant underpricing.³ This suggests that asymmetry of information about demand does not play a central role in IPOs.

Several papers have tried to explain underpricing as the product of signaling – examples are Allen and Faulhaber (1989), Grinblatt and Hwang (1989), and Welch (1989).⁴ In these models, issuing firms have private information about their value and try to signal the quality of their prospects to outside investors through the offering price. The rationale for underpricing is that good firms prefer to "leave money on the table" when going public. This should credibly signal the issuing firm's type and allow the issuer to profit from future equity issues. A problem with this literature is that it predicts a relationship between first day returns and subsequent seasoned equity issues that is not found in the data, as shown by Michaely and Shaw (1994). Furthermore, all these models assign a passive role to financial intermediaries. This seems at odds with the fact that hardly any issuing firm directly places its shares in the market.

Similarly to these contributions, our model explains underpricing as a result of asymmetric information. However, in our analysis underpricing emerges as the result of the conflict of interest between the privately informed issuer and the intermediary. As such, our explanation for underpricing does not rely on a "leaving money on the table" argument and is independent of seasoned equity offerings.

Finally, a further strand of literature on IPOs deals with the reasons to go public. Important contributions include Zingales (1995), Chemmanur and Fulghieri (1999), Mello and Parsons (2000), Brau et al. (2003), Faure-Grimaud and Gromb (2004). Although this is not the focus of our analysis, our modelling approach is consistent with several of the reasons

³These IPOs are characterized by the presence of underwriting syndicates. Consistent with our story, institutions whose incentives are not perfectly aligned with those of the issuer participate in the underwriting and distribution of shares. Muscarella and Vetsuypens (1989) also report that in these IPOs the maximum offering price is decided by an independent underwriter. This could be interpreted as a way to avoid oversignaling.

⁴Alternative or complementary explanations for underpricing are surveyed by Ritter (2003). Important contributions have focused on: information asymmetries among investors (Rock, 1986) information acquisition by the investment bank (Benveniste and Spindt, 1989), prospect theory (Loughran and Ritter, 2002), lawsuit avoidance (Hughes and Thakor, 1992), IPOs as marketing events (Chemmanur 1993, Aggarwal et al. 1992, Demers and Lewellen 2003, Habib and Ljungqvist, 2001).

highlighted by the literature.

3 Model Background

The model is introduced in section 4. Here, we discuss two main features of our modelling approach.

Recent empirical evidence (see Brau and Fawcett 2006) suggests that one of the main reasons why firm owners may be reluctant to go public is their desire to retain ownership and/or control. Intuitively, the benefits from retaining ownership/control should increase with the firm's quality. Consistent with this idea, a key feature of our analysis is that:

(a) Owners of high quality firms are more reluctant to go public than owners of low quality firms.

The point can be easily illustrated by analogy with a standard lemon setting. Consider a seller who wishes to sell a good, in exchange for a payment p from the buyer. The good may be either of type H (high quality) or of type L (low quality). The seller's valuation for a type q = H, L good is v_q . Since type H goods have greater value, it is natural to assume that

$$v_H > v_L \tag{1}$$

Condition (1) implies that, when contemplating trade, the outside option of the owner of a high quality good – namely, keeping the good – is higher than that of the owner of a low quality good. In turn, this makes owners of high quality goods more reluctant to sell. Assumption (a) applies this notion to the case of firms undertaking an IPO. Owners of high quality firms are more reluctant to go public in the sense that any payment that would induce them to go public would also induce owners of low quality firms to do so – but not vice versa.

There is no general agreement on why firms go public.⁵ However, there is some consensus (see Ritter and Welch 2002) that

- (i) a firm's owner desire to finance further investments/growth opportunities within the firm and/or;
- (ii) his desire to liquidate his position in the firm (cashing out) in order to finance new ventures

constitute important reasons. Brau and Fawcett (2006), in a survey of chief financial officers, find strong support for (i) and moderate support for (ii).⁶ In section A.1 of the appendix,

⁵Alternative explanations for why firms go public comprise: i) to facilitate acquisitions (Brau et al. 2003), ii) to establish a market price for the firm (Zingales 1995, Mello and Parsons 2000, Faure-Grimaud and Gromb 2004), iii) to obtain ownership dispersion (Chemmanur and Fulghieri 1999).

⁶More precisely, they report that more than 30% of CFOs felt that the IPO provides a chance to cash out for the principal and/or for the venture capitalist.

we formally show that assumption (a) is compatible with both classes of situations, (i) and (ii).⁷ In both cases, the issuer accepts to share (or forgo) the return of some assets he owns in order to raise cash. The issuer's outside option to going public is therefore determined by the quality of the assets already in place. Consistent with Tirole (2006, pp. 245-246), the higher the value of these assets, the more reluctant the issuer is to go public. The model we consider thus differs from existing signaling models of IPO underpricing, in that we assume that the outside option of the issuer depends on his type.⁸ (In what follows, we will use the expressions "owner of a type q firm" and "type q issuer" interchangeably.)

Another feature of our setup is that

(b) Going public is socially efficient (inefficient) if the firm is of type H (type L).

Consider again a standard lemon setting. Suppose that there is a single buyer, and the maximum payment he is willing to make for a type q good is u_q . When a type q is sold, value is created so long as $u_q > v_q$, i.e. the maximum payment that the buyer is willing to make is above the minimum payment that the seller would accept. The requirement in (b) can be thus summarized by the following:

$$u_H > v_H, \quad u_L < v_L \tag{2}$$

In a frictionless world, going public never destroys value. In reality, however, there are compelling reasons why going public may entail costs that private companies do not face. A typical example is the cost of complying with a more stringent regulation. For instance, a survey by CRA international, a consultant, finds that the cost of complying with section 404 of the Sarbane Oxley act ranges between 1.5 billion dollars for small companies to 7.5 billions for large companies. Alternative examples include the costs resulting from executives having to spend time negotiating with shareholders and regulators, rather than "getting things done". For going public to be efficient, the potential gain from trade must be sufficiently high to outweigh such costs. In our model, this happens only for type H issuers.

In section A.1 of the appendix, we discuss how examples (i) and (ii) are compatible with assumption (b) when (a) is also satisfied. The key factor is that the value of the assets in place is positively correlated with the value of the investment opportunity to be financed. A natural explanation for this correlation is the persistency of issuer-specific factors such as

⁷It should however be stressed that other theories of going public are not necessarily in conflict with our assumption. For instance, in Chemmanur and Fulghieri (1999), the alternative to going public for a firm consists of raising finance privately with a venture capitalist. In equilibrium, high quality firms obtain better conditions from the venture capitalist. Hence, when considering whether to go public, high quality firms have a higher outside option.

⁸Typical models of IPO underpricing assume that the issuer seeks to raise cash to finance a project by offering a share of the future cash flows to investors. While projects differ in their expected value, issuers of both types have identical outside options. This implies that issuers of type L are more reluctant to go public than issuers of type H. Whatever share of future cash flows an issuer of type L would be willing to forgo, a type H issuer would also be willing to forgo. This in turn allows type H issuers to reveal their type by "leaving money on the table", i.e. by offering a relatively high share of future cash flows to investors. A stylized version of these types of models is discussed in Tirole 2006, pp. 262-264.

entrepreneurial ability, human capital, business or political connections. Issuers with more valuable investment opportunities are thus more reluctant to go public because they have more valuable assets in place.

It should be noted, however, that, while assumption (b) allows us to derive our results in a particularly striking form (especially in section 5, where the issuer tries to market the shares directly to the investors), it is not the driving force of our story. This point is discussed further in section 8.

4 The model

We consider issues in the primary market through fixed price offers. There is an issuer (S), an investor (I), and an investment bank (B). We compare two possible mechanisms for issuing stocks: 1) direct issues, 2) intermediated issues. In a direct issue, S chooses the offering price, I decides whether to buy, and B retains a passive role. In an intermediated issue, B acts as an underwriter. She is the only counterparty for both S and I, and can bargain with S over the offering price.

For simplicity, we concentrate on a model in which the offering price is the only choice variable. This naturally arises in IPOs where either the amount of cash that the issuer wishes to raise or the number of shares on offer are determined by exogenous forces.⁹ As discussed in section 8, this assumption does not play a crucial role for our results.

4.1 Payoffs

In this section, we formally introduce and discuss payoffs of the issuer and of the investor. Section A.1 of the appendix illustrates how these payoffs may emerge within the context of the examples (i) and (ii) mentioned in section 3 – namely financing further growth and cashing out.

Issuer

The issuer S is risk neutral. S's firm can be of two types: $q \in \{H, L\}$ where H indicates a high quality firm, while L denotes a low quality one. We assume that S's type is private information to S.

The issuer's payoff from going public net of his outside option is V(p,q), where p is the offering price and $q \in \{L, H\}$. V(., q) is assumed to be continuous and differentiable, and to satisfy the following monotonicity conditions:

A 1.

⁹The first case emerges when for instance the investment opportunity to be financed through the IPO is characterized by indivisibilities. The second case applies for instance to privatization IPOs, where the share of the firm that remains in public hands is fixed by regulators. More generally, as argued by Biais, Bossaerts and Rochet (2002), in IPOs "[..] the number of shares is indeed most of the time set a priori".

(i) For all $p, p' \in \mathbb{R}^+$, p > p' and $q \in \{L, H\}$

$$V(p,q) > V(p',q) \tag{3}$$

(ii) For each $q \in \{L, H\}$, there exists $v_q \in \mathbb{R}^+$ such that

$$V(v_q, q) = 0 (4)$$

(iii) For all $p \in \mathbb{R}^+$

$$V(p,H) < V(p,L) \tag{5}$$

A 2. For all p and p' such that $p > p' \ge v_H$, $\frac{V(p,H)}{V(p,L)} > \frac{V(p',H)}{V(p',L)}$.

Assumption A1(i) states that the benefit from going public to the issuer increases in the price at which shares are sold to the investors. The intuition is straightforward if the issuer aims at cashing out by going public. However, the assumption stands even if the purpose of the IPO is to raise finance to be invested in the firm. Intuitively, keeping the amount of finance raised constant, a higher offering price implies that the issuer retains a larger stake in the firm. He will therefore be able to claim a larger share of the firm's future cash flows.

Assumption A1(ii) guarantees the existence of reservation prices (v_H and v_L for type H and L respectively). A type q issuer would never choose to go public if the offering price were below v_q . Assumption A1(iii) ensures that a type L issuer would profit more from going public than a type H issuer. Notice that assumptions A1(ii) and A1(iii) imply $v_H > v_L$, namely condition (a) described in section 3.

Assumption A2 provides a sorting condition. It implies that whenever type L weakly prefers the highest among two offering prices, type H strictly prefers the highest. Formally, let $p \geq v_H$ and p' < p be two offering prices and let x and x' denote the probabilities that the IPO is successful at p and p' respectively. Then A2 implies that xV(p, H) > x'V(p', H) whenever $xV(p, L) \geq x'V(p', L)$. Intuitively, type H benefits relatively more than type L from a higher price even when this reduces the chances of selling the shares.

Investor

The investor I is risk neutral. We denote I's net payoff from investing in the firm through the IPO with U(p,q), where U(.,q) is continuous and differentiable. The restrictions on U(p,q) are symmetric to those in A1.

A 3.

(i) For all $p, p' \in \mathbb{R}^+$, p > p' and $q \in \{L, H\}$

$$U(p,q) < U(p',q) \tag{6}$$

(ii) For each $q \in \{L, H\}$, there exists $u_q \in \mathbb{R}^+$ such that

$$U(u_q, q) = 0 (7)$$

(iii) For all
$$p \in \mathbb{R}^+$$

$$U(p,H) > U(p,L)$$
 (8)

Assumption A3(i) ensures that I's net payoff is decreasing in the offering price. Assumption A3(ii) guarantees the existence of I's reservation prices for type H and L (u_H and u_L) respectively. With perfect information, I would accept to buy shares in a type q firm only if the offering price were less than u_q . Assumption A3(iii) ensures that I prefers type H to type L. Finally, as in the case of the issuer, assumption A3(ii) combined with assumption A3(iii) imply $u_H > u_L$.

A simple example of payoffs that satisfy assumptions A1-A3 is the linear case $V(p,q) = p - v_q$ and $U(p,q) = u_q - p$. This corresponds to a standard lemon model where the value of the firm is v_q for the issuer and u_q for the investor.

We concentrate on situations where the adverse selection problem is particularly severe, in that, on efficiency grounds, low quality firms should not go public at all. This is condition (b) discussed in the previous section. Accordingly, we make the following assumption:

A 4. The surplus generated when a type $q \in \{H, L\}$ firm goes public

$$V(p,q) + U(p,q) \tag{9}$$

is independent of p and is positive for q = H and negative for q = L.

Note that assumption A4 implies that $u_H > v_H$ and $u_L < v_L$. The requirement that the surplus generated by trade be independent of the offering price is natural if we interpret p as a mere transfer of wealth from the investor to the issuer. This is for instance the case in the simple linear example sketched above. More generally, A4 allows us to establish a clear benchmark under which to evaluate direct and intermediated issues. This is because social welfare is only affected by trade and not by the prices at which trade occurs.

Investment Bank

The Investment Bank B performs an active role only in intermediated issues, which are discussed in section 6. We postpone the discussion of the presentation of B's payoff to that section.

4.2 Information structure

The issuer perfectly observes the firm's type. The information structure of other agents is as follows.

Investor. I's prior is that S is of type H with probability λ and of type L with probability $1 - \lambda$. Prior beliefs are common knowledge among all players. I observes two signals: the offering price, p, and an exogenous private noisy signal, $s \in [\underline{s}, \overline{s}]$, with conditional density f(s|q) and cumulative distribution F(s|q).

We assume that f(s|q) satisfies the Monotone Likelihood Ratio Property (MLRP) so that $\frac{f(s|H)}{f(s|L)}$ is a strictly increasing function of s and has full support $(0, \infty)$.

Investment Bank. We assume that B receives a signal $\sigma \in \{h, l\}$ about S's type; σ is not observed by I and is distributed as follows

$$\Pr(\sigma = h|H) = \eta \tag{10}$$

$$\Pr(\sigma = h|L) = 1 - \eta \tag{11}$$

for $\eta \in (1/2, 1)$.¹⁰ B's posteriors $\pi_h \equiv \Pr(H|h)$ and $\pi_l \equiv \Pr(H|l)$, with $\pi_h > \pi_l$, are thus

$$\pi_h = \frac{\eta \lambda}{\eta \lambda + (1 - \eta)(1 - \lambda)} \tag{12}$$

$$\pi_l = \frac{(1-\eta)\lambda}{(1-\eta)\lambda + \eta(1-\lambda)} \tag{13}$$

5 Direct Issues

The timing of a direct issue is as follows:

Stage 0 Nature draws $q \in \{H, L\}$ from a Bernoulli distribution with $Pr(q = H) = \lambda$.

Stage 1 S observes q and selects an offering price $p \in \mathbb{R}^+$.

Stage 2 I observes p, his private signal $s \in [\underline{s}, \overline{s}]$ and chooses whether to buy or not.

Stage 3 payoffs are realized.

If no trade occurs at stage 2, then both S and I obtain their outside options.

The game just described is a signaling game between S and I and the appropriate equilibrium concept is Perfect Bayesian Equilibrium (PBE). Denote with $\mu(q|p,s)$ the belief function giving I's probability assessment that S is of type q given p and s. A PBE is a strategy profile for S and I and a belief function $\mu^*(q|p,s)$ which satisfy the usual conditions: 1) S's best reply, 2) I's best reply, 3) consistency of $\mu^*(q|p,s)$ with Bayes rule for all p that are selected with positive probability in equilibrium. In order to avoid the common "unsent message" problem, we refine the PBE concept with Cho and Kreps (1987) version of "Never a Weak Best Response" (NWBR). Intuitively, for any p that is selected with probability zero in equilibrium, if the set of I's best responses for which a type q issuer weakly benefits from selecting p (relative to his equilibrium payoff) is contained in the set for which a type q'

The cases in which the intermediary is perfectly informed $(\eta = 1)$ or uninformed $(\eta = 1/2)$ are qualitatively equivalent. These are discussed in section 8.

issuer strictly benefits, then I, upon observing p, assigns probability zero to type q. This a standard refinement in the signaling literature. Section 8 analyzes the robustness of our results to the use of this equilibrium concept.

We are now ready to state the main result of the section.

Proposition 1. Under direct issue, there exists a unique NWBR-refined equilibrium outcome and is such that S charges some price $p \ge u_H$ and no trade occurs.

The proof relies on two observations. The first is that there is no separating equilibrium in which trade occurs. Consider an equilibrium in which a type q issuer selects action p_q with $p_L \neq p_H$, and trade occurs with positive probability. In this equilibrium, type L would not be trading since, when q = L, trade would make either the issuer or the investor worse off. This follows from the assumption that it is socially inefficient to trade type L firms (S's reservation price v_L is greater than I's reservation price u_L). However, if type H were trading, this equilibrium would violate type L's incentive compatibility.

The second observation is that other types of equilibria in which trade may occur (pooling or hybrid) would violate NWBR. Whenever both types of issuer are pooled together at the same price, say p^* , a type L issuer would benefit from trading more than a type H issuer, since $v_H > v_L$. This implies that the set of I's best responses for which type L weakly benefits from a deviation, $p > p^*$, is contained in the set of best responses for which type H strictly benefits. According to NWBR, I's beliefs should then assign probability zero to the event that a type L issuer deviated to p. This gives the issuer a strong incentive to raise p in order to signal that he is of type H and hence increase the likelihood of trading.

A perhaps more intuitive way to explain the result of proposition 1 is the following. If offering prices were perfectly revealing, type L issuers would want to increase their price to mimic type H issuers. If offering prices were not perfectly revealing, type H issuers would want to raise their price to differentiate themselves from type L. This "upward race" would only stop when the offering price hit the investor's reservation utility for a type H firm. At that price, I never chooses to buy the shares as long as the price is selected by type L with positive probability. Hence, the market breaks down.

Although of similar flavor, the result of proposition 1 is thus different from the classic example of Akerlof (1970). In Akerlof's case the market breaks down because adverse selection exerts a downward pressure on the price. In our case, the reverse happens. The market breaks down because signaling concerns exert an upward pressure on the price. In a sense, there is "over-signaling". If S could ex-ante commit not to use the offering price as a signal for his type, he would always be able to trade with a positive probability. This is because, conditional on his private signal being sufficiently high, the investor would be willing to buy at any pooling price that does not exceed his reservation utility. Hence, rather than solving the adverse selection problem, signaling through prices exacerbates it.

¹¹See for instance Fudenberg and Tirole (1991, pp. 454-56).

6 Intermediated Issues

In the previous section we have seen how trade collapses when S tries to market his shares directly to I. In this section, we show that this inefficiency can be mitigated by the presence of an investment bank acting as an underwriter. We identify conditions under which such an intermediary can restore trade between S and I and avoid losing money in the process.

As an underwriter, B buys all shares from the issuer and resells them to the investor. (In what follows, we use the expressions "the IPO takes place" and "B underwrites the shares" interchangeably, to signify that trade occurs between B and S.) For this service, B receives from S a fixed compensation, denoted with ϕ . The transfer ϕ is essentially an underwriting fee which is paid to B in exchange for her underwriting services. The payment of ϕ is therefore contingent on the IPO taking place. We do not explicitly model the market for underwriting services, but rather assume that B takes ϕ as given. The notion that underwriters do not condition their compensation on the characteristics of the IPO but instead stick to a given fee is backed by recent evidence by Chen and Ritter (2000). They show that, for the US, in more than 90% of IPOs raising between 20 and 80 million dollars, the underwriter compensation was exactly 7% of the value of the issue.

The offering price is determined after B observes her signal σ about S's type. An aspect in which this model differs from existing literature is that B is able to influence the offering price. In order to avoid the complications that naturally arise when considering bargaining under asymmetric information, we assume an extreme form of bargaining in which B makes a take it or leave it offer to S about the offering price. If S accepts the price offered by B, then the IPO takes place. In this case S makes the transfer ϕ to B and B announces the offering price to the investor. S's net payoff is thus $V(p,q) - \phi$. For future reference we denote with v_H^{ϕ} the offering price such that

$$V(v_H^{\phi}, q) - \phi = 0 \tag{14}$$

If S rejects B's offer, then no IPO takes place and all players obtain their outside option. For simplicity, we assume that in this case S cannot use a different underwriter. ¹²

The timing of an intermediated issue is thus as follows.

Stage 0 the market for underwriting services determines ϕ .

Stage 1 Nature draws a type $q \in \{H, L\}$ for S which is observed by S only.

Stage 2 B observes σ and makes an offer p to S.

Stage 3 S chooses whether to accept of reject B's offer.

If the offer is rejected the game ends and players obtain their outside options. If it is accepted,

¹²Although not explicitly modelled here, one can suppose that the investor interprets using a different underwriter after the underwriter has acquired information as a signal that the information about the issuer is unfavorable.

Stage 4 B announces p to I.

Stage 5 I observes p and s and chooses whether to buy or not.

Stage 6 payoffs are realized.

A feature of our setup is that it rules out any form of contracting on the offering price prior to stage 2. As discussed by Ellis et al. (1999), this is consistent with typical IPO procedures. The firm and the underwriter generally meet to choose the offering price only on the day prior to the placement of the stocks. By that time, the process of information collection by the underwriter has already taken place.

Given B's role as underwriter, it is natural to assume that B acts as a self-interested agent with incentives that may be different from those of other players. We assume that when the IPO does not take place, B obtains a payoff equal to zero. When the IPO takes place and I chooses to buy, shares are transferred from S to I and B's payoff is equal to ϕ . If, by converse, I chooses not to buy, the shares remain in the hands of B. In this case, we need to determine the utility that B derives from holding a stake in the firm. A natural starting point is to assume that there are gains from trade between B and I – namely that holding shares is more valuable to I than to B. If this were not the case, then it would be unclear why B should act as an intermediary rather than being an investor. On the other hand, if both have access to the stock market, then the return that B and I can realize from owning shares in a type q = H, L firm should be the same. For instance, I may want to buy shares in order to resell them on the stock market at a later date, when the firm's type has been observed. If this can be replicated by B, then their returns should coincide. If both have access to stock market, the gains from trade between B and I should then arise from different opportunity costs. 13 This is the route we take here. We assume that the net payoffs of B and I are identical up to a constant capturing the difference in opportunity costs. B's net payoff from a type q firm when I chooses not to buy is therefore $U(p,q) - K + \phi$, where the constant $K \geq 0$ represents the gains from trade between B and I.

A strategy for B is a map from the set of realizations of σ into the set of probability distributions over \mathbb{R}^+ (i.e. the set of admissible values for p). A strategy for S is a map from $\{H, L\} \times \mathbb{R}^+$ (i.e. the set of realizations of q and the set of possible p offered by B) into the set of probability distributions over {accept, reject}. Finally, a strategy for I is a map from $\mathbb{R}^+ \times [\underline{s}, \overline{s}]$ into the set {buy, not buy}. A PBE is a profile of strategies and belief functions for B, I, and S such that at any stage of the game: 1) strategies are optimal given beliefs, 2) beliefs are consistent with Bayes rule for all actions played with positive probability in equilibrium. An equilibrium for the intermediated case is then a PBE of the game just described and a level of ϕ such that B makes non-negative expected profits. Again, we focus on equilibria that pass NWBR.

From assumption A1(iii), $V(p, L) - \phi > V(p, H) - \phi$ for all ϕ and $p \in \mathbb{R}^+$. This implies that, whenever a type H is willing to accept B's offer, a type L issuer would also accept.

¹³Different opportunity costs may for instance arise if the investment bank has investment opportunities not available to the average investor, as in Sherman (1999).

An offering price is therefore either accepted by both types or only by type L. B is thus unable to weed type L out of the market by appropriately selecting the offering price. On the other hand, before choosing the price, B observes a signal that is not observed by I. Hence, although the choice of the offering price cannot perfectly reveal the issuer's type, it may nevertheless convey information about the realization of B's signal.

The remainder of this section is organized as follows. We first provide two results that illustrate the tension that emerges between the intermediary's ability to restore trade and her viability. We then turn to the full characterization of the benchmark case in which B's expected profits are exactly zero, showing that this is compatible with trade.

6.1 Trade vs Viability

In section 5, we saw how the issuer's signaling concern led to market breakdown. In this section we analyze the trade off between the requirements that trade should occur and that the intermediary should break even. We start off by considering the intermediary's pricing strategy.

If the offering price is too low to be accepted by type H, but is accepted by type L, then B would surely lose from underwriting the shares. Intuitively, this follows from the assumption that no gain from trade can be reaped from trading type L shares. Hence, B may find it optimal to go ahead with the IPO only at a price p that is acceptable to a type H issuer, i.e. at a price greater than or equal to v_H^{ϕ} . Conditional on $p \geq v_H^{\phi}$, I believes that issuer's type is H with a positive probability. If the offering price is also lower than his valuation for type H shares, u_H , I is willing to buy the shares whenever his signal s is sufficiently high. Hence, trade between B and I occurs with positive probability. The next lemma shows that B would always find it optimal to select a price lower than u_H whenever this is compatible with type H's participation. Hence, once B has chosen to underwrite the shares, the "no trade" equilibrium identified in proposition 1 can no longer emerge.

Lemma 1. Assume that there is trade between B and S (i.e. the IPO takes place). Then, $v_H^{\phi} < u_H$ is both necessary and sufficient to ensure that trade between B and I occurs with positive probability.

The proof relies on the following argument. Suppose that B chooses a price $p \geq u_H$ so that I never buys. At this price, shares are on average overpriced. B would certainly gain by paying a lower price to the issuer. Hence, B has an incentive to decrease price below u_H . But then, trade with I occurs with positive probability.

Lemma 1 highlights how B's incentives on price setting differ from those of the issuer. Intuitively, although B is a seller when dealing with I, she is a buyer when dealing with S. As a buyer, B suffers from a higher price – since, given ϕ , her payoff from owning the shares, U(p,q)-K, is decreasing in p. In contrast, the issuer's payoff V(p,q) is increasing in p. This difference in B and S's price-setting incentives implies that, when the intermediary

¹⁴This intuition is formally proved by lemma C.2 in the Appendix.

is present, the upward pressure on price that characterizes direct issues is mitigated. Trade may therefore occur at equilibrium.

This discussion stresses the desirability of a conflict of interest between the issuer and the intermediary. This is in contrast with the existing literature on intermediaries on IPOs, which has mainly focused on the design of mechanisms geared at aligning the two parties' incentives.¹⁵

The natural next question is whether B can gain from underwriting the shares. Since B has imperfect information, I's choice of buying or not conveys information about the value of the shares. Hence, whenever I chooses to buy at the offering price, B upwardly revises her valuation for the shares. Similarly, whenever I chooses not to buy at the offering price, B should revise her valuation downwards. Essentially, B faces an adverse selection problem, or "seller's curse". As a result, B on average makes losses from her underwriting activities. For B to break even, it is therefore necessary that ϕ , the underwriting fee, be strictly positive. This shares similarities with a well known result in the literature on market microstructure. Glosten and Milgrom (1985) show that, in the presence of informed traders, a risk neutral dealer would need a positive bid-ask spread in order to break even. The result is illustrated by the following lemma.

Lemma 2. In any equilibrium in which the IPO takes place, B makes no expected losses only if $\phi > 0$. If the IPO takes place at price p, the following conditions are satisfied: $p \geq v_H^{\phi} > v_H > v_L > u_L$.

As lemma 2 illustrates, the underwriting fee plays an essential role, which is that of compensating B of the expected losses from underwriting.

The requirement that $\phi > 0$ has implications for the price at which the IPO may occur. As seen above, a type H issuer is willing to accept B's offering price only if this is greater than v_H^{ϕ} , which is increasing in ϕ . Intuitively, the larger the fee the issuer must pay, the higher the offering price must be in order to induce him to sell the shares. Hence, the higher the expected losses from underwriting, the higher the offering price.

Lemma 1 and lemma 2 highlight a key tradeoff. On the one hand, the presence of an intermediary may allow trade by mitigating the upward pressure on price that is present with direct issues. On the other hand, the requirement that this intermediary should be viable introduces a different sort of upward pressure on the price. Formally, this tension is illustrated by the requirements imposed on ϕ by the two lemmata. From lemma 1, trade between B and I requires v_H^{ϕ} , and therefore ϕ , to be sufficiently small. However, lemma 2 suggests that the intermediary is viable only when ϕ is sufficiently large. These two requirements are in conflict with each other. The main question is therefore whether viability and trade may coexist. We show that this is indeed possible, by characterizing the benchmark case in which B makes exactly zero profit. Recent evidence suggests that this case is also empirically relevant. Indeed, as shown in Hansen (2001), the typical 7% spread earned by investment banks is too low to yield abnormal profits.

¹⁵See for instance Baron (1982), Hölmstrom and Baron (1980), and Biais, Bossaerts, and Rochet (2002).

6.2 Characterization and existence of the zero profit equilibrium

In this section we restrict attention to the case where B makes exactly zero profit. We divide the analysis in two parts. First, we characterize the zero-profit equilibrium. We then discuss the sufficient conditions for the equilibrium to exist, and for it to be unique.

Lemma 3. (Characterization of the zero-profit equilibrium) Let Φ denote the set of values of ϕ for which trade between B and I occurs with positive probability and B makes zero profits. Whenever $\phi \in \Phi$, a NWBR-refined equilibrium with trade exhibits perfect separation. The IPO takes place only when B observes $\sigma = h$, in which case she offers the lowest price satisfying the participation constraint of type H (i.e. v_H^{ϕ}). When $\sigma = l$ is observed, B offers a price that violates the participation constraint of both types of issuers and no IPO takes place.

Lemma 3 establishes several results. First, in a zero profit equilibrium the IPO takes place only when B receives favorable information about S's type ($\sigma = h$). By contrast, when information is unfavorable ($\sigma = l$), B proposes a price so low that the offer is always rejected by S, and no IPO takes place. Upon receiving unfavorable information, B therefore prefers to forgo the IPO altogether. Lemma 3 thus highlights how the presence of an intermediary may allow separating equilibria to emerge. As seen in section 5, when the issuer sells his shares directly to the investor, mimicking behavior by type L issuers would systematically destroy any separating equilibrium. By contrast, when B observes $\sigma = l$, her incentive to pretend otherwise is much weaker. So long as I trades on the basis of his private information, the likelihood of being unable to sell overpriced shares is relatively high if B underwrites them when $\sigma = l$. This result suggests that the intermediary acts as a screening device, ensuring that firms that manage to go public have, on average, higher quality than those which fail to do so. This benefits I, since B's choice to underwrite or not credibly reveals her information. In a sense, through underwriting, B is forced to "put her money where her mouth is".

Second, when the IPO takes place, B selects the lowest price at which the high quality issuer is willing to sell the shares. A higher price would benefit the issuer, but would hurt the intermediary. By setting a higher price, B would sell the shares with a lower probability since I would only buy for higher realizations of his signal. As seen in the discussion of lemma 2, B revises her valuation for the shares downwards when I chooses not to buy them. Paying a higher price to the issuer and selling with a lower probability is thus unambiguously detrimental to B. Therefore, conditional on type H's participation, the intermediary's expected profits are strictly decreasing in p. Lemma 3 thus shows how B's pricing strategy in the zero-profit equilibrium totally diverges from S's pricing strategy under direct issues: B selects the lowest possible price that satisfies type H's participation constraint.

The next lemma provides sufficient conditions for existence and uniqueness of a zero profit equilibrium.

¹⁶On the other hand, mimicking would allow B to profit from the underwriting fee. However, it turns out that the value of ϕ ensuring that B makes zero profit when $\sigma = h$ is low enough to discourage mimicking when $\sigma = l$.

Lemma 4. (Existence and uniqueness of a zero-profit equilibrium) If $(1 - \pi_h)U(u_H, L) + V(u_H, H) - K > 0$, then Φ is non-empty. If U(p, H) - U(p, L) is non-increasing in p, Φ is a singleton for K sufficiently small.

The condition $(1 - \pi_h)U(u_H, L) + V(u_H, H) - K > 0$ ensures the existence of a fee such that B makes zero profits and trade between B and I occurs with positive probability. The expression $(1 - \pi_h)U(u_H, L) - K$ represents the maximum expected loss that the intermediary may suffer from underwriting.¹⁷ The term $V(u_H, H)$ corresponds to the maximum fee that is compatible with participation by a high quality issuer. When this is greater than the maximum expected loss, a positive probability of trade is assured.

Note that, given $u_H > v_H > u_L$, $V(u_H, H) > 0$ and $U(u_H, L) < 0$. Hence, provided that K is not too large, existence is ensured for sufficiently high values of π_h . This makes clear that our results do not rely on B having inferior information to S. Indeed, trade is most likely to occur when B's information is almost perfect. Note that π_h is increasing both in η and in λ . This can be used to make predictions on the conditions that favor trade while ensuring the viability of the intermediary. If type L are frequently drawn (low λ), B's information should be very precise (high η). In other words, when bad firms are frequent, B must be an effective screening device. By converse, if B's information is not precise, type L firms must be infrequent. The requirement that K should not be too large has a natural interpretation. In equilibrium, I only buys when his signal s is sufficiently high. Hence, B stands to keep the shares with a positive probability. If K is too large, then the value of ϕ required to compensate B from losses incurred in that instance would be so large as to outweigh any gain from trade between I and S.

Finally, the requirements for uniqueness ensure that the marginal gain from quality to the buyer (U(p, H) - U(p, L)) does not increase with the price. This is however not necessary for the existence of the equilibrium.

The following proposition summarizes the results discussed in this section.

Proposition 2. When B makes zero expected profits, trade from S to I may occur with positive probability provided that B's information is not too imprecise and/or K is not too large. Under these conditions, B goes ahead with the IPO only when her signal is favorable, in which case she selects the lowest price at which the type H issuer is willing to trade.

7 Implications

We now discuss the implications of the model for underpricing. The objective is to assess whether shares that change hands are on average under or overpriced. We define as underpricing (overpricing) a situation in which, at the offering price, the investor would make a profit (loss) if buying.¹⁸ Average underpricing then occurs if, at the equilibrium price, the expected quality on offer is such that the investor would make expected profits.

¹⁷When the price is equal to u_H , I never buys. Hence, the expected loss from underwriting is $\pi_h U(u_H, H) + (1 - \pi_h)U(u_H, L) - K = (1 - \pi_h)U(u_H, L) - K$.

 $^{^{18}}$ Using B rather than I as a benchmark for defining under/overpricing would not change any of the results.

We show that the causes of underpricing are twofold. First, the intermediary suffers from an informational disadvantage vis-à-vis the investor. This implies that shares bought by I are on average underpriced while shares bought by B are on average overpriced. Second, the pricing behavior of the intermediary results in shares being on average underpriced (independently of who buys them) even when B's informational disadvantage becomes vanishingly small.

Proposition 3 summarizes the first point.

Proposition 3. Assume that B's expected profits are zero and trade between B and I occurs with positive probability (i.e. $\phi \in \Phi$). Then, in equilibrium: i) shares in the hands of I are on average underpriced; ii) shares in the hands of B are on average overpriced.

This result is a direct consequence of the adverse selection problem suffered by B when trading with I. In the zero profit equilibrium this problem is especially pronounced since B's pricing strategy perfectly reveals her information to I. By contrast, I's information remains private. Hence, I has an informational advantage vis-à-vis B. The seller's curse is therefore extreme. On average, when B manages to sell the shares, these are underpriced; when she is unable to sell them, they are overpriced.

The result has an implication for the relationship between underpricing and amount subscribed. In equilibrium, shares of both type L and type H issuers are underwritten by the investment bank with positive probability. Those of type H issuers are underpriced while those of type L issuers are overpriced. I is more likely to observe a high realization of his private signal s when the issuer is of type H than when the issuer is of type L. Hence, he is more likely to buy when the issuer is of type H. This implies that there is a positive correlation between I's decision to buy and the likelihood that shares are underpriced. This is line with the findings of Cornelli and Goldreich (2003) who find positive correlation between underpricing and oversubscription.

To fully appreciate the role of the intermediary, it is important to assess whether average underpricing emerges simply as a result of the intermediary's informational disadvantage or it is also driven by her pricing incentives. To this purpose, we consider what happens when B's informational disadvantage becomes vanishingly small. We thus analyze two limiting cases: i) B is almost perfectly informed, and ii) I is almost uninformed. Notice that B is almost perfectly informed when $\pi_h \to 1$. I is almost uninformed when the signal he receives converges to a random variable that is independent of q. Let then $\tilde{s} \in (\underline{s}, \overline{s})$ be a random variable with distribution \tilde{F} independent of q.

Proposition 4. Assume that ϕ is such that B's expected profits are zero. For $s \rightarrow {}^d \tilde{s}$, the probability of trade between B and I converges to a positive value if and only if, at the equilibrium price v_H^{ϕ} , shares are on average underpriced. For $\pi_h \rightarrow 1$, trade between B and I occurs with positive probability if and only if shares are on average underpriced.

Proposition 4 shows that, in the limiting cases, trade occurs if and only if shares are on average underpriced. The intuition relies on the positive relationship between the equilibrium price and B's informational disadvantage vis-à-vis I. The greater this informational

disadvantage, the more severe the seller's curse, and the greater the fee necessary to cover B's expected losses from her underwriting activities. In order to ensure S's participation, B then has to charge a higher price. Otherwise, S would not accept to trade, given that he has to pay a large fee to B. By converse, when B's signal is sufficiently precise, or I's signal is sufficiently imprecise, the fee necessary for B to break even is small. The price charged by B is accordingly low and underpricing occurs.

8 Robustness and Extensions

In this section, we informally discuss how our results may apply also when some of our assumptions do not hold. We concentrate on four issues. First, we consider a situation where the issuer may potentially signal his type through his choice of both a share price and the number of shares he puts up for sale on the market. Second, we discuss possible modifications of the information structure. As a third point, we address the implications of ignoring equilibrium refinements, and allowing multiple equilibria to emerge in the model. Finally, we consider a situation where, under perfect information, both high and low quality firms should be traded.

As will become clear below, all cases are characterized by a common theme. The key idea is that under direct issue, low quality firms have a strong incentive to mimic high quality firms. This reduces either the average quality or the amount of trade that can be achieved in equilibrium. By contrast, the intermediary's incentive to mimic when she has received unfavorable information are comparatively weak. As a result, distortions are reduced.

Most of the claims we make in this section are formally proved in section D of the appendix.

8.1 Allowing for two instruments

The model analyzed in previous sections assumes that the only instrument available to the issuer to signal his type is the share price. What would happen if the issuer could vary both the share price and the number of shares on sale? In that case, he would have two rather than one instrument at his disposal. It is therefore legitimate to wonder whether this greater scope for manoeuvre could allow the high quality issuer to credibly signal himself, eliminating the problems highlighted in Section 5. We argue that this conjecture is misguided. Under conditions (a) and (b) of Section 3, mimicking behavior by low quality firms may not be prevented, even if the issuer can use both price and number of shares as signals. Intuitively, any combination (price, number of shares) such that the high quality issuer wishes to undertake the IPO would also attract the low quality issuer. This point can be illustrated using the simple linear payoffs framework. Suppose that the value of S's firm is v_q . By going public, S sells a fraction 1-z of the firm for a payment p. The issuer's net payoff from going public is:

$$V(p, z, q) = p - (1 - z)v_q \tag{15}$$

Since $v_H > v_L$, it follows that V(p, z, H) < V(p, z, L) for all $z \in (0, 1)$ and p > 0. In this case, any pair (p, z) that satisfies type H's participation constraint would also satisfy type L's.

Consider now type L's incentive compatibility. In a separating equilibrium, type L would be unable to trade (given $u_L < v_L$). It follows that he would always profit from mimicking type H. Separating equilibria are therefore not possible.

8.2 Information Structure

The information structure introduced in section 4 may raise a number of concerns: (1) Does it matter that the issuer has better information than the investment bank? (2) What would happen if the investment bank had no private information? We argue that our results are robust to these modifications of the information structure.

To address question (1) suppose that both the issuer and the intermediary perfectly observe the issuer's type. Since the intermediary is perfectly informed, she no longer suffers from an informational disadvantage vis-à-vis the investor. As a result, $\phi > 0$ is no longer required for her to break even. It can be shown that there exist a continuum of zero-profit equilibria such that ϕ is zero and the IPO takes place only when the issuer is of type H, in which case the investor buys with probability one. However, contrary to the result in lemma 3, the equilibrium price is no longer uniquely determined under NWBR, but can take any value in $[v_H, u_H]$. Underpricing still emerges in all but one of the possible equilibria.

Now consider question (2). If the intermediary possesses no private information, her pricing decisions do not convey any information. The intermediary's incentives, however, are unchanged. In the zero-profit equilibrium, her expected payoff is strictly decreasing in the offering price. Hence, she selects the lowest price acceptable to a type H issuer. However, here the informational disadvantage the intermediary suffers vis-à-vis the investor is maximal, and the value of ϕ required for her to break even is correspondingly large.

The argument sketched above makes clear that our results are qualitatively independent of the precision of the intermediary's information. Although having a well-informed intermediary may be desirable, this is not a necessary condition for intermediaries playing an important role in the market.

8.3 Direct Issues are Viable

As suggested by proposition 1, without intermediaries, the IPO market would collapse. Throughout the paper, we have relied on this result to justify the existence of intermediaries in IPOs. Here, we extend the analysis to cases in which proposition 1 may not fully apply so that direct issues could be viable. One may wonder whether there is any need for intermediaries in these cases. We argue that, in most circumstances, the presence of an intermediary increases either the amount or the average quality of trade (or both). This could provide a possible justification for the existence of intermediaries even when issuing shares directly would be possible.

8.3.1 Equilibria that fail NWBR

In the main body of the paper we use an equilibrium refinement (NWBR) that dramatically reduces indeterminacy. The refinement ensures that, when considering direct issues, the

unique (refined) equilibrium involves no trade at all, and that, under intermediated issues, the offering price is unique when the intermediary's profits are zero. An alternative approach is that of ignoring the refinement and allowing for multiple equilibria. We argue that this would not invalidate our results. To see this, consider first the case of direct issues. There exist a continuum of perfect Bayesian equilibria such that both types are pooled at some offering price $p \in [v_H, u_H]$. In these equilibria, I uses a threshold strategy on his signal when observing p and chooses not to buy when observing any other price. This strategy is in turn sustained by (NWBR-failing) beliefs assigning probability one to type L when observing any price different from p. Efficiency would require that H firms be traded with probability one and L firms with probability zero. These equilibria are thus inefficient, since low quality firms may be traded with a positive probability and high quality firms are traded with probability less than one.

Consider now intermediated issues. As noted in 8.2, if the intermediary is perfectly informed, then a situation where the IPO takes place only when the issuer is of type H (in which case the investor buys with probability one) is an equilibrium. In this equilibrium, the intermediary achieves full efficiency, something that could not be reached through direct issues. Hence, the use of an intermediary is clearly beneficial. What if the intermediary is only imperfectly informed? Without refinement, the equilibrium price in the zero-profit equilibrium is not necessarily $p_h = v_H^{\phi}$. However, perfect Bayesian equilibria still involve full revelation of the intermediary's information through the price choice (the intermediary goes on with IPO only when she receives favorable information). This stands in contrast with the case of direct issues in which no information is revealed. However, it also introduces a different source of inefficiency. If the intermediary has received misleading information, she may prevent the investor from trading with a high quality issuer. A trade-off then arises. On the one hand, in the separating equilibrium, the intermediary reveals her information to the investor. Keeping everything else equal, this improves efficiency. On the other hand, the intermediary may also mistakenly prevent good firms from going public. As her information becomes more precise, the second effect weakens, while the first becomes stronger. When the intermediary is perfectly informed, the second effect disappears altogether.

8.3.2 Low quality firms should also be traded

A running hypothesis of our model is that low quality firms would not be traded under perfect information. This assumption plays an important role in ensuring that, under direct issues, no separating equilibrium is possible, and trade may collapse altogether. A natural question is therefore whether the case for intermediaries would collapse if all firms generated gains from trade when going public independently of their quality. We argue that this is not the case. As discussed in Ellingsen (1997), in this case the only refined equilibrium that may emerge with direct issues is a separating equilibrium where trade is rationed for high quality firms – i.e., high quality firms sell their shares with probability less than one. By contrast efficiency would require that all types of firms be traded with probability one.

Consider now intermediated issues. First, suppose that the intermediary is perfectly informed. In this case, there exists a continuum of equilibria in which trade occurs for sure.

We refer to these equilibria as "efficient equilibria" since they maximize social welfare. In an efficient equilibrium, the intermediary selects

$$p = \begin{cases} p_H \in [v_H, u_H] \text{ when } q = H\\ p_L \in [v_L, u_L] \text{ when } q = L \end{cases}$$
 (16)

the issuer always chooses to sell at p_q , q = H, L, and the investor buys. Efficient equilibria are typically separating (i.e. $p_H \neq p_L$) – although a pooling can also be efficient when $u_L > v_H$. These equilibria are sustained, for instance, by beliefs assigning probability one to type H when observing any out of equilibrium price in the interval $[v_H, u_H]$ and probability one to type L for prices lower than v_H . It is easy to check that these beliefs pass NWBR. Given the other party's strategy, both the investor and the intermediary have no incentive to deviate. It is then immediate to check that selling at p_q is a best reply for S. When one of these equilibria is selected, the presence of an intermediary unambiguously improves welfare, since it ensures full efficiency.

Now suppose that the intermediary is not perfectly informed. As seen in lemma 3, the intermediary then selects the price to maximize the probability of selling the shares she has underwritten. Suppose that $u_L > v_H$. Then, if $v_H^{\phi} < u_L$, by selecting a price $p \in \left[v_H^{\phi}, u_L\right]$ the intermediary would be able to sell the shares with probability one. This is clearly her favorite course of action. Intuitively, therefore, the zero-profit equilibrium outcome would have: $\phi = 0$, $p \in [v_H, u_L]$. This outcome would again generate full efficiency, since trade between the issuer and the investor would occur with certainty. Overall, therefore, the case where the intermediary is fully informed and the case where $u_L > v_H$ provide examples of how the use of an intermediary may be desirable, even when both high and low quality firms should be traded.

9 Concluding Remarks

This paper provides a possible rationale for the presence of financial intermediaries in IPOs, an issue that has been largely ignored by previous theoretical literature. We have shown how, in certain circumstances, signaling concerns by issuers may cause the market to break down when intermediaries are absent. The presence of a price-setting intermediary acting as an underwriter restores trade. However, the intermediary is not financially viable unless the underwriting fee she receives from the issuer is sufficiently high. This is potentially problematic, since hefty fees may dissuade issuers from going public, even when this would be efficient. Nonetheless, we show that a zero-profit equilibrium – where the intermediary just breaks even on average – can exist. In this equilibrium, the intermediary acts as a screening device, by agreeing to underwrite only the shares of firms over which she has favorable information. Consistent with empirical evidence, our model predicts that whenever the IPO takes place, shares subscribed by investors are underpriced.

Our model opens up several avenues for future research. For instance, it would be interesting to provide an explicit analysis of the market for intermediary services. In principle,

competition in this market may take two different forms. On the one hand, we may have intermediaries competing to attract firms wishing to go public. On the other hand, we may have different firm-intermediary pairs competing to attract investors. Whether fully unregulated competition would deliver trade is not entirely clear. Our paper has shown that, for trade to occur, a conflict of interests must exist between the issuer and the intermediary. However – at least in the first case described above – in order to become more attractive to potential clients, intermediaries may have an incentive to find devices that align their interests with those of the issuers. So, in the absence of any form of regulation, market forces could potentially act against efficiency. Interventions that limit the firms' scope for manoeuvre in aligning their interests with the issuers' could then enhance efficiency.

A Appendix

A.1 Model Background

In this section we provide a rationale for the assumptions on the payoffs. We sketch two stories. In the first the issuer seeks cash to finance an expansion of the firm's activities. In the second a venture capitalist wants to cash out part of the value of his current venture in order to invest in a new venture. Several elements are common to both examples:

- All players can invest at the market rate $r \geq 1$.
- The issuer cannot borrow.
- There is a small cost in going public: a small amount c > 0 of resources is wasted in order to complain with regulatory requirements (transparency etc.).
- The issuer has already invested an amount a > 0 in the firm.
- Only type H firms yield above market returns. The per unit of finance return of the initial investment a is $R_H > r$ for firms of type H and is $R_L = r$ for firms of type L.

For both stories, we discuss the assumptions that ensure that conditions (a) and (b) discussed in section 2 are met. We also provide simple numerical examples that illustrate how, under (a) and (b), all the other assumptions we make are met rather naturally.

(i) Financing Further Growth. Consider the case of an entrepreneur who relies on the stock market to finance a possible expansion of his firm. There are two types of entrepreneurs: high ability (type H) and low ability (type L). Firms of type H entrepreneurs are high quality firms, in that they have growth opportunities (positive NPV projects), while firms of type L entrepreneurs do not. More precisely, entrepreneurs of type H can make further investments with positive NPV. We assume for simplicity that these further investment opportunities consist of one project to be conducted within the firm, which requires one unit of finance. If less than one unit of finance is invested, the project is unsuccessful, and yields a zero return. Provided that the unit of finance is invested, the project is successful and yields an above market return. We let this return be denoted as $R_H^{IPO} > r$. Following Tirole (2006, p. 244), we assume that it is not possible to contract on the cash flow generated by this additional project separately from that of the projects already in place within the firm. Moreover, since the entrepreneur is credit-constrained, we restrict attention to situations where the IPO allows him to raise the whole unit of finance he requires.

Type L firms have no positive NPV projects, but only carry projects that yield the market rate r. Hence, they have the same unit return whether they go public and raise finance or stay private: $R_L^{IPO} = R_L = r$.

Given R_H the return of assets in place for q = H, the present value of the firm to the entrepreneur in the absence of IPO is

$$\begin{cases}
\frac{R_H a}{r} & if \quad q = H. \\
a & if \quad q = L.
\end{cases}$$
(A.1)

This represents the opportunity cost incurred by the entrepreneur when going public. Note that, since $R_H > r$, this opportunity cost is always greater when q = H than when q = L. This reflects the persistence of entrepreneurial ability: not only do type H entrepreneurs have better investment opportunities, they also have more valuable firms.

The value of the firm after the IPO has taken place (and one extra unit is injected in the firm) is

$$\begin{cases}
\frac{R_H^{IPO} + R_H a}{r} - c & if \quad q = H. \\
1 + a - c & if \quad q = L.
\end{cases}$$
(A.2)

The investor's alternative to purchasing the shares is that of investing his unit of capital at the market rate. The net surplus generated when the firm goes public is thus equal to

$$\begin{cases}
\frac{R_H^{IPO} - r}{r} - c & if \quad q = H. \\
-c & if \quad q = L.
\end{cases}$$
(A.3)

Given c > 0, it is clear that, from an efficiency standpoint, low quality firms should not go public (condition (b)). This is because, when quality is low, going public entails no benefit (since the cash raised is used to finance a project that yields the same return as the market), but only costs. In contrast, provided that c is not too large – so that $\left(R_H^{IPO} - r\right)/r - c > 0$ – high quality firms should indeed go public. By going public, the entrepreneur is able to finance a project that yields returns exceeding those provided by the market.

We now turn to condition (a). This is satisfied if a type L entrepreneur would be willing to go public for all offering prices such that a type H entrepreneur would be willing to do so, but not vice versa. Suppose that the entrepreneur offers a fraction 1-z of the company's profits in exchange for one unit of finance to be injected into the company. Let the total number of shares be normalized to one. The price of a share is thus $p = \frac{1}{1-z}$ so that $z = 1 - \frac{1}{p}$ (henceforth denoted as z(p)). The net payoff of a type q = H, L entrepreneur is:

$$V(p,q) = \begin{cases} z(p) \left(\frac{R_H^{IPO} + R_H a}{r} - c \right) - \frac{R_H a}{r} & if \quad q = H. \\ z(p) (1 + a - c) - a & if \quad q = L. \end{cases}$$
(A.4)

while the net payoff for the investor is:

$$U(p,q) = \begin{cases} (1-z(p)) \left(\frac{R_H^{IPO} + R_H a}{r} - c\right) - 1 & if \quad q = H. \\ (1-z(p)) (1+a-c) - 1 & if \quad q = L. \end{cases}$$
(A.5)

The conditions that need to be satisfied to induce the entrepreneur to go public are

$$z(p) \ge \begin{cases} \frac{R_H a}{R_H^{IPO} + R_H a - cr} & if \quad q = H. \\ \frac{a}{1 + a - c} & if \quad q = L. \end{cases}$$
 (A.6)

Condition (a) is satisfied whenever

$$c < \frac{R_H - R_H^{IPO}}{R_H - r} \tag{A.7}$$

Condition (A.7) ensures that the minimum share price at which type H issuers are willing to go public is higher than for type L. This is always the case whenever c is not too large and

$$R_H > R_H^{IPO} \tag{A.8}$$

Inequality (A.8) states that there are decreasing returns to investment. For instance, the (financially constrained) issuer may have allocated the initial a to the project with the highest NPV. Further projects, while still ensuring a positive NPV, will yield a lower return.¹⁹

It is straightforward to verify that all our restrictions are satisfied for reasonable parameter values. Consider for instance a = r = 1, $R_H^{IPO} = 1.25$, $R_H = 1.5$. In this case, we have

$$V(p,q) = \begin{cases} z(p) (2.75 - c) - 1.5 & \text{if } q = H. \\ z(p)(2 - c) - 1 & \text{if } q = L. \end{cases}$$

$$U(p,q) = \begin{cases} (1 - z(p)) (2.75 - c) - 1 & \text{if } q = H. \\ (1 - z(p)) (2 - c) - 1 & \text{if } q = L. \end{cases}$$
(A.9)

so that
$$v_H = \frac{2.75-c}{1.25-c}$$
, $v_L = \frac{2-c}{1-c}$, $u_H = 2.75-c$ and $u_L = 2-c$

The requirement for condition (b) to be met – namely, that $\left(R_H^{IPO}-r\right)/r-c>0$ – becomes: $c<\frac{1}{4}$. Whenever this is the case, $u_H>v_H>v_L>u_L$ and assumptions A2-A4 are satisfied. As for A1, the requirement that $V(p,H)< V(p,L) \ \forall p\in \mathbb{R}^+$ is unnecessarily restrictive, and was imposed in section 4 only for notational convenience. Even if $V(p,H)\geq V(p,L)$ for some $p>u_H$, this is irrelevant since these prices violate I's participation constraint. The relevant requirement is therefore that $V(p,H)< V(p,L) \ \forall p\in [0,u_H]$. It is straightforward to verify that, in the numerical example, V(p,H)< V(p,L) for all p such that z(p)<2/3. This is always met for all values of $p\in [0,u_H]$.

¹⁹This is in line with the findings of Pagano, Panetta, and Zingales (1998) who document that profitability decreases after the IPO.

decreases after the IPO. ^20 For instance, $\frac{d(V(p,H)/V(p,L))}{dp} = \frac{1}{4} \frac{1-2c}{(z(p)(c-2)+1)^2} \frac{dz}{dp} > 0$ given c < 1/4.

(ii) Cashing out. Consider now the case of a venture capitalist (VC) who wants to raise one unit of finance to be invested in a new venture. The setup is very similar to that of case (i). The only difference here is that the new investment is not carried out within the existing firm but within a new venture whose cash flows are entirely appropriated by the VC. As discussed above and in section 3, the types H and L can be interpreted as capturing the VC's ability (or experience). A more skilled VC has greater ability to identify profitable projects. The profitability of both his current and new ventures is therefore higher.

The notation is the same as in case (i) with the exception of R_q^{IPO} which now indicates the return that the VC obtains from investing the unit of capital in the new venture. As before, we set $R_L^{IPO}=R_L=r$.

The outside option for the VC when going public is given by (A.1). Expression (A.2) now represents the present value of combined assets from the existing firm and the new project when the IPO takes place. The net surplus generated by the IPO is given by (A.3). Therefore, as in the previous case, type L firms should never go public whereas type H firms should go public if $c < (R_H^{IPO} - r)/r$. When this holds, condition (b) is satisfied.

We now turn to condition (a). For simplicity we impose a = 1. The VC's net payoff is:

$$V(p,q) = \begin{cases} \frac{R_H^{IPO}}{r} + z(p) \left(\frac{R_H}{r} - c\right) - \frac{R_H}{r} & if \ q = H. \\ z(p) (1 - c) & if \ q = L. \end{cases}$$
(A.10)

while the net payoff for the investor is:

$$U(p,q) = \begin{cases} (1-z(p))\left(\frac{R_H}{r} - c\right) - 1 & if \ q = H. \\ (1-z(p))(1-c) - 1 & if \ q = L. \end{cases}$$
(A.11)

The conditions that need to be satisfied to induce the VC to go public are

$$\begin{cases}
z(p)(R_H - cr) \ge R_H - R_H^{IPO} & if \quad q = H. \\
z(p)(1 - c) \ge 0 & if \quad q = L.
\end{cases}$$
(A.12)

If c < 1, type L goes public for all $z(p) \ge 0$. In that case, condition (a) is satisfied if $R_H > R_H^{IPO}$. If $c \ge 1$, type L never goes public. Condition (a) is thus never satisfied – since, at best, both types are equally reluctant to undertake the IPO. Overall, therefore, the necessary and sufficient conditions for (a) are:

$$c < 1 \tag{A.13}$$

and

$$R_H > R_H^{IPO} \tag{A.14}$$

The first requirement is straightforward. The second is equivalent to condition (A.8). Again, this is consistent with the idea of a VC selecting first the projects with higher NPV.

It is straightforward to verify that all our restrictions are satisfied using the same parameter values as in example (i): r = 1, $R_H^{IPO} = 1.25$, $R_H = 1.5$. We have

$$V(p,q) = \begin{cases} z(p) (1.5-c) - 0.25 & \text{if } q = H. \\ z(p)(1-c) & \text{if } q = L. \end{cases}$$

$$U(p,q) = \begin{cases} (1-z(p)) (1.5-c) - 1 & \text{if } q = H. \\ (1-z(p)) (1-c) - 1 & \text{if } q = L. \end{cases}$$
(A.15)

so that
$$v_H = \frac{1.5-c}{1.25-c}$$
, $v_L = 0$ (for $c < 1$), $u_H = 1.5-c$ and $u_L = 1-c$

The requirement for condition (b) to be met – namely, that $\left(R_H^{IPO}-r\right)/r-c>0$ – becomes: $c<\frac{1}{4}$. Whenever this is the case, $u_H>v_H>v_L>u_L$ and assumptions A2-A4 are satisfied. As for A1, the discussion at the end of example (i) applies. It is straightforward to verify that V(p,H)< V(p,L) for all p such that z(p)<1/2. This is always met for all values of $p\in[0,u_H]$.

B Proof of Proposition 1

We start by showing that there exists no separating equilibrium in which trade occurs. Then we show that no pooling or hybrid equilibrium in which trade occurs passes NWBR. Finally, we show that there exists a NWBR-refined equilibrium in which no trade occurs.

Lemma B.1. There is no separating equilibrium in which trade occurs.

Proof. In a separating equilibrium, I always discards her private signal as equilibrium prices are fully informative. Let \mathcal{P}_q be the set of p selected in equilibrium by type q. If $\mathcal{P}_L \cap \mathcal{P}_H = \emptyset$, type L is never able to trade since $u_L > v_L$. However, type L would benefit from trading at any $p \in \mathcal{P}_H$ given that p is optimal for type H and $v_L < v_H$. Hence, type L would always try to mimic type H. \square

Lemma B.2. No pooling-hybrid equilibrium in which trade occurs survives NWBR.

Proof. Assume that trade occurs in a pooling or hybrid equilibrium. Suppose that pooling occurs at \hat{p} , with $v_H \leq \hat{p} < u_H$. A type H issuer selects \hat{p} with probability $\beta_H \in (0, 1]$ and a type L Issuer announces \hat{p} with probability $\beta_L \in (0, 1]$. I observes \hat{p} and receives a signal s. I's expected net payoff from buying at \hat{p} is:

$$\frac{\lambda \beta_H f_H(s)}{\lambda \beta_H f_H(s) + (1 - \lambda) \beta_L f_L(s)} U(\hat{p}, H) + \frac{(1 - \lambda) \beta_L f_L(s)}{\lambda \beta_H f_H(s) + (1 - \lambda) \beta_L f_L(s)} U(\hat{p}, L)$$
(B.1)

²¹For instance, $\frac{d(V(p,H)/V(p,L))}{dp} = \frac{0.25(1-c)}{(z(p)(1-c))^2} \frac{dz}{dp} > 0$ given c < 1/4.

Expected utility is nonnegative if:

$$\frac{f_H(s)}{f_L(s)} \ge -\frac{(1-\lambda)\beta_L}{\lambda\beta_H} \frac{U(\hat{p}, L)}{U(\hat{p}, H)}$$
(B.2)

Notice that the LHS is an increasing function of s and the RHS is positive for $\hat{p} \in (u_L, u_H)$. Given the full support assumption, there always exists a threshold $s^* \in [\underline{s}, \overline{s}]$ such that (B.2) holds if $s \geq s^*$ and does not hold if $s < s^*$. Hence, I's threshold strategy is to buy if $s \geq s^*$ and not to buy for $s < s^*$. S's payoff is:

$$[1 - F_q(s^*)]V(\hat{p}, q)$$
 (B.3)

where $q \in \{H, L\}$. Suppose now that I observes a deviation $p > \hat{p}$. Upon observing p, I uses a threshold s^D (see Bénabou and Tirole 2003 on this way to use NWBR). According to NWBR, type L can be eliminated from the deviation if the set of values for s^D that make him weakly benefit from the deviation is contained in the set of values that make type H strictly benefit. Type L would (weakly) benefit whenever:

$$[1 - F_L(s^D)]V(p, L) \ge [1 - F_L(s^*)]V(\hat{p}, L)$$
(B.4)

Type L is eliminated if, whenever (B.4) holds, the following also holds:

$$[1 - F_H(s^D)]V(p, H) > [1 - F_H(s^*)]V(\hat{p}, H)$$
 (B.5)

Note that (B.5) is always verified whenever $s^D \leq s^*$ since the issuer would get a higher price and a lower threshold (which implies a higher probability to sell). Consider then $s^D > s^*$. For a deviation $p > \hat{p}$, assumption A2 implies that (B.5) is always satisfied when (B.4) holds so long as:

$$\frac{1 - F_H(s^D)}{1 - F_H(s^*)} \ge \frac{1 - F_L(s^D)}{1 - F_L(s^*)}$$
(B.6)

Rewrite the above as:

$$(1 - F_H(s^D))(1 - F_L(s^*)) - (1 - F_H(s^*))(1 - F_L(s^D)) \ge 0$$
(B.7)

The derivative of the above expression with respect to s^D is

$$-f_H(s^D)(1 - F_L(s^*)) + f_L(s^D)(1 - F_H(s^*))$$
(B.8)

so that the LHS of equation (B.7) is increasing whenever:

$$\frac{f_H(s^D)}{f_L(s^D)} < \frac{1 - F_L(s^*)}{1 - F_H(s^*)} \tag{B.9}$$

and is decreasing whenever the reverse inequality holds. Given the MLRP (which implies that $\frac{f_H(s^D)}{f_L(s^D)}$ is an increasing function), the LHS of inequality (B.7) must be an increasing-decreasing function (i.e. increasing for small values of s^D and decreasing beyond a threshold).

We note that the limits of (B.7) for $s^D \to s^*$ and $s^D \to \overline{s}$ are both zero. Since the LHS of inequality (B.7) is an increasing-decreasing function which converges to zero as s^D moves toward the bounds of (s^*, \overline{s}) , it follows that it cannot be negative in (s^*, \overline{s}) . Hence, (B.6) holds and type L can be always eliminated. Since type L can be eliminated, for deviations to $p < u_H$, I would always buy with probability one. But then, it is always optimal for S to deviate to $p \in (\hat{p}, u_H)$, which implies that there cannot be any pooling or hybrid equilibrium with trade. \square

Lemma B.3. There always exists a NWBR-refined equilibrium in which trade does not occur.

Proof. Consider a situation in which S always announces $p=u_H$ and I selects a threshold equal to \overline{s} for all p. This is clearly an equilibrium if I believes any deviation to emanate from type L. It is also robust to NWBR since, for any deviation $p \geq v_H$, the set of I's best responses that make type L willing to deviate coincides with the set of best responses that make type H willing to deviate. Therefore, type L cannot be eliminated. \square

C Intermediated Issues

We start by establishing a number of intermediate results that will be extensively used to prove the results in sections 6 and 7. Lemmata C.1-C.2 provide some characterization of the prices that may emerge in any equilibrium with trade. Lemma C.3 focuses on I's best reply in stage 5, taking ϕ and p as given. Lemma C.4 focuses on B's interim payoff given ϕ . We then turn to the proofs of the results stated in sections 6 and 7.

Lemma C.1. Trade between B and I occurs only if $p < u_H$.

Proof. For any $p \geq u_H$, I would always lose from trading unless p were exactly equal to u_H and I knew the issuer to be of type H for sure. This however cannot happen since: i) given assumption A1, type L would be willing to trade at $p = u_H$ whenever type H would be willing to trade, ii) neither B nor I are able to perfectly discriminate between L and H, given the information at their disposal. \square

Lemma C.2. If B cannot make losses, the IPO takes place only if $V(p, H) - \phi \ge 0$.

Proof. The IPO can take place only if type H is willing to sell his shares. This follows from the assumption that gains from trade are positive only when the firm's quality is high. Suppose that the IPO takes place and only type L is willing to sell. Then, if trade between B and I occurs, I's payoff is U(p, L), type L's payoff is $V(p, L) - \phi$, and B's payoff is equal to ϕ . Given assumption A4, the sum of all payoffs is negative for all p. Hence, someone would be better off by not participating. Suppose now that there is no trade between B and I. Then, B's payoff is $U(p, L) - K + \phi$ and I's payoff is zero. Again, the sum of B and S's payoffs is negative, implying that either B or S would be better off by not participating. \square

As mentioned in section 6, for a given $\phi, \ v_H^{\phi}$ is the price level solving:

$$V(v_H^{\phi}, H) - \phi = 0 \tag{C.1}$$

so that lemma C.2 can be equivalently expressed as $p \geq v_H^{\phi}$.

We now turn to I's optimal strategy at stage 5. The next lemma shows that, abstracting from B's incentive to participate, the necessary conditions in lemmata C.1 and C.2, are sufficient for trade between S and I.

Lemma C.3. Assume that $v_H^{\phi} < u_H$ and that B offers a price $p \ge v_H^{\phi}$. Then, I follows a threshold strategy $s^*(p)$ on his signal s. $s^*(p)$ satisfies:

$$\begin{cases}
s^* = \overline{s} & p \ge u_H \\
\lambda f_H(s^*) \Pr(p|H) U(p,H) + (1-\lambda) f_L(s^*) \Pr(p|L) U(p,L) = 0 & u_L
(C.2)$$

where Pr(p|q), $q \in \{H, L\}$ denotes the probability that I ascribes to observing p given type q.

Corollary C.1. Trade between B and I occurs with positive probability at any p such that $v_H^{\phi} \leq p < u_H$.

Proof. Recall that B's strategy is a map from the set $\{h,l\}$ of realizations of her signal σ to the set of probability distributions over p. Since σ is, conditionally on q, independent of s, p is also independent of s conditionally on q. Hence, it is easy to show that:

$$Pr(q|s, p) = \frac{f_q(s) Pr(p|q) Pr(q)}{\sum_{q \in \{H, L\}} f_q(s) Pr(p|q) Pr(q)}$$
(C.3)

I's expected payoff from buying at p is therefore:

$$\frac{\lambda f_H(s) \Pr(p|H) U(p,H) + (1-\lambda) f_L(s) \Pr(p|L) U(p,L)}{\lambda f_H(s) \Pr(p|H) + (1-\lambda) f_L(s) \Pr(p|L)}$$
(C.4)

This can also be written as

$$\frac{\lambda\left(\frac{f_{H}(s)}{f_{L}(s)}\right)\Pr(p|H)}{\lambda\left(\frac{f_{H}(s)}{f_{L}(s)}\right)\Pr(p|H) + (1-\lambda)\Pr(p|L)}U(p,H) + \frac{(1-\lambda)\Pr(p|L)}{\lambda\left(\frac{f_{H}(s)}{f_{L}(s)}\right)\Pr(p|H) + (1-\lambda)\Pr(p|L)}U(p,L)$$
(C.5)

Given $p \ge v_H^{\phi}$, $\Pr(p|H) > 0$ and $\Pr(p|L) > 0$ (We assume that S accepts to trade when indifferent). The derivative of (C.5) with respect to s is

$$\frac{d\left(\frac{f_H(s)}{f_L(s)}\right)}{ds} \frac{\lambda \left(1-\lambda\right) \Pr(p|H) \Pr(p|L)}{\left(\lambda \left(\frac{f_H(s)}{f_L(s)}\right) \Pr(p|H) + \left(1-\lambda\right) \Pr(p|L)\right)^2} \left[U(p,H) - U(p,L)\right] \tag{C.6}$$

From the MLRP, $\frac{f_H(s)}{f_L(s)}$ is a strictly increasing function of s. From assumption A3, U(p,H)-U(p,L)>0. Hence, (C.6) is positive, implying that (C.5) is strictly increasing in s. Therefore, I follows a threshold strategy. Namely, there exists a value $s^*(p)$ such that, for $s \leq s^*(p)$, I does not purchase the shares, while, for $s > s^*(p)$, I purchases the shares.

For $p \geq u_H$, $U(p, L) < U(p, H) \leq 0$. Hence, (C.5) is negative for all s and, therefore, $s^*(p) = \overline{s}$ (no trade between B and I). For $p \leq u_L$, $U(p, H) > U(p, L) \geq 0$. Hence, (C.5) is positive for all s and, therefore, $s^*(p) = \underline{s}$ (trade between B and I occurs with probability one).

Given $u_L , <math>U(p, H) > 0$ and U(p, L) < 0. For $s \to \overline{s}$, $f_H(s)/f_L(s) \to +\infty$. U(p, H) > 0 then implies that (C.5) is positive. For $s \to \underline{s}$, $f_H(s)/f_L(s) \to 0$. Given U(p, L) < 0, (C.5) is negative. By monotonicity and continuity, there exists a unique value s^* such that (C.5) is equal to zero. Finally, setting (C.5) equal to zero and rearranging yields the expression in (C.2). \square

We can now derive B's expected payoff in the subgame starting in stage 2 from announcing a price at which S is willing to trade.

Lemma C.4. Denote as p_{σ} the price set by B upon observing $\sigma \in \{h, l\}$, and as $s^*(p_{\sigma})$ I's threshold when observing p_{σ} . B's interim expected payoff when the IPO takes place is:

$$\pi_{\sigma} F_{H}(s^{*}(p_{\sigma})) \left(U(p_{\sigma}, H) - K \right) + (1 - \pi_{\sigma}) F_{L}(s^{*}(p_{\sigma})) \left(U(p_{\sigma}, L) - K \right) + \phi$$
 (C.7)

Proof. Given σ and I's threshold strategy s^* , the conditional probability that I does not buy and S is of type H is:

$$\Pr(H, s < s^* | \sigma) = \frac{\Pr(s < s^*, \sigma | H) \Pr(H)}{\Pr(\sigma | H) \Pr(H) + \Pr(\sigma | L) \Pr(L)} =$$

$$= \frac{\Pr(\sigma | H) \Pr(H)}{\Pr(\sigma | H) \Pr(H) + \Pr(\sigma | L) \Pr(L)} \Pr(s < s^* | H) = \pi_{\sigma} F_H(s^*)$$
(C.8)

for $\sigma \in \{h, l\}$. By the same token, $\Pr(L, s < s^* | \sigma) = (1 - \pi_{\sigma}) F_L(s^*)$. Expression (C.7) follows. \square

C.1 Proof of Lemma 1

According to lemmata C.1, C.2, and C.3, $v_H^{\phi} \leq p < u_H$ is necessary and sufficient for trade between B and I to occur with positive probability. We now show that B has incentive to charge such a price when $v_H^{\phi} < u_H$. If the IPO takes place, then lemma C.2 implies that B must be charging $p \geq v_H^{\phi}$. Hence, all we need to show is that B has incentive to charge $p < u_H$. Suppose then that B has received a signal $\sigma \in \{h, l\}$ and that, at equilibrium, she charges $p_{\sigma} \geq u_H$ so that no trade occurs between her and I. Given that U(., L) is decreasing in p and $U(p, H) \leq 0$ for $p \geq u_H$, B's expected payoff is at most

$$(1 - \pi_{\sigma})U(u_H, L) - K + \phi \tag{C.9}$$

By deviating, and charging a lower price $v_H^{\phi} \leq p' < u_H$, B could sell with a positive probability. Denoting as s' I's threshold in that case, B's expected payoff would be

$$\pi_{\sigma} F_H(s') \left(U(p', H) - K \right) + (1 - \pi_{\sigma}) F_L(s') \left(U(p', L) - K \right) + \phi$$
 (C.10)

Now,

$$\pi_{\sigma} F_{H}(s') \left(U(p', H) - K \right) + (1 - \pi_{\sigma}) F_{L}(s') \left(U(p', L) - K \right) + \phi >$$

$$> (1 - \pi_{\sigma}) U(u_{H}, L) - K + \phi$$
(C.11)

if

$$\pi_{\sigma} F_{H}(s') U(p', H) + (1 - \pi_{\sigma}) \left[F_{L}(s') U(p', L) - U(u_{H}, L) \right] + K \left[1 - \pi_{\sigma} F_{H}(s') - (1 - \pi_{\sigma}) F_{L}(s') \right] > 0$$
 (C.12)

Notice that, since $p' < u_H$, U(p',H) > 0. Moreover, since U(.,L) is strictly decreasing and $U(u_H,L) < 0$, $F(s' \mid L)U(p',L)-U(u_H,L) > 0$. Finally, $1-\pi_{\sigma}F_H(s')-(1-\pi_{\sigma})F_L(s') \geq 0$. Hence, the inequality is always satisfied. By charging $v_H^{\phi} \leq p' < u_H$, B is strictly better off than by charging $p_{\sigma} \geq u_H$. \square

C.2 Proof of Lemma 2

Assume $\phi \leq 0$. Two cases may arise: a) $p \leq u_L$ and b) $p > u_L$. Consider case a). For $p \leq u_L$, I is willing to buy for all realizations of s. Hence, trade between B and I occurs with probability one, so that B's net payoff is equal to ϕ . If $\phi < 0$, B makes expected losses. If $\phi = 0$, lemma C.2 shows that the IPO takes place only if $p \geq v_H^{\phi} = v_H > u_L$, which contradicts $p \leq u_L$. Consider now case b). Assume first $p \geq u_H$ so that no trade occurs between B and I. In this case, B's profits are at most:

$$(1 - \pi_{\sigma})U(u_H, L) - K + \phi$$
 (C.13)

Given $U(u_H, L) < 0$ and $K \ge 0$, B's profits are negative for all $\phi \le 0$. Assume now $p < u_H$ so that trade between B and I occurs with positive probability.

Given lemma C.3, I follows a threshold strategy $s^*(p)$ such that:

$$\lambda \Pr(p|H) f_H(s) U(p,H) + (1-\lambda) \Pr(p|L) f_L(s) U(p,L) < 0 \tag{C.14}$$

for all $s < s^*(p)$. Notice that:

$$\Pr(p|q) = \begin{cases} \eta \beta_h + (1-\eta)\beta_l & q = H\\ (1-\eta)\beta_h + \eta \beta_l & q = L \end{cases}$$
 (C.15)

where $\beta_{\sigma} \equiv \Pr(p|\sigma)$ is derived from B's equilibrium strategy (we omit the argument p, but it should be clear that β_{σ} is a function of p). Inequality (C.14) can be thus rewritten as:

$$\lambda [\eta \beta_h + (1 - \eta)\beta_l] f_H(s) U(p, H) + (1 - \lambda)[(1 - \eta)\beta_h + \eta \beta_l] f_L(s) U(p, L) < 0$$
 (C.16)

Since the inequality holds for all $s \leq s^*(p)$, one can integrate between \underline{s} and $s^*(p)$ to obtain

$$\lambda[\beta_h \eta + \beta_l (1 - \eta)] F_H(s^*(p)) U(p, H) +$$

$$+ (1 - \lambda)[\beta_h (1 - \eta) + \beta_l \eta] F_L(s^*(p)) U(p, L) < 0$$
(C.17)

When B follows a strategy that consists of announcing p with probability β_{σ} upon observing σ , B's ex-ante payoff is:

$$\sum_{p \in \mathcal{P}} \{ \lambda [\beta_h \eta + \beta_l (1 - \eta)] F_H(s^*(p)) [U(p, H) - K] + (1 - \lambda) [\beta_h (1 - \eta) + \beta_l \eta] F_L(s^*(p)) [U(p, L) - K] + \Gamma \phi \}$$
(C.18)

where $\Gamma \equiv \lambda[\beta_h \eta + \beta_l (1-\eta)] + (1-\lambda)[\beta_h (1-\eta) + \beta_l \eta] > 0$ and \mathcal{P} denotes the set of prices announced with positive probability. Given $K \geq 0$ and (C.17), B's expected profits for any $p \in (u_L, u_H)$ can be non-negative only if $\phi > 0$. This proves the first statement of lemma 2. The second statement follows from the first statement $(v_H^{\phi} > v_H)$ and lemma C.2 $(p \geq v_H^{\phi})$.

C.3 Proof of Lemma 3

In order to prove lemma 3, we need to characterize the equilibrium in the subgame starting in stage 2. We first discuss B's interim participation constraint. This is used to show that the IPO takes place if and only if B observes $\sigma = h$, so that no trade between B and S occurs when B observes $\sigma = l$. We then show that there is only one equilibrium with trade that passes NWBR, and this is such that $p_h = v_H^{\phi}$.

Once σ is observed, the price p_{σ} must satisfy B's interim participation constraint. Otherwise, B could offer a price p_{σ} so low that S would always reject it and make zero profits – a situation de facto equivalent to no IPO occurring at all. Moreover, in the candidate equilibrium, B's expected profits prior to observing σ must be zero. This can only happen if the interim participation constraint is satisfied with equality.

B's interim payoff is derived in lemma C.4. If the IPO takes place for $\sigma = h$, the price p_h must then satisfy:

$$\pi_h F_H(s^*(p_h)) \left(U(p_h, H) - K \right) + (1 - \pi_h) F_L(s^*(p_h)) \left(U(p_h, L) - K \right) + \phi = 0 \tag{C.19}$$

Similarly, if the IPO takes place when $\sigma = l$, p_l satisfies:

$$\pi_l F_H(s^*(p_l)) \left(U(p_l, H) - K \right) + (1 - \pi_l) F_L(s^*(p_l)) \left(U(p_l, L) - K \right) + \phi = 0$$
 (C.20)

Lemma C.5. The IPO takes place only when $\sigma = h$.

Proof. We proceed by contradiction. Consider an equilibrium in which the IPO takes place when B observes $\sigma = l$. In equilibrium, the incentive compatibility of B when observing $\sigma = h$ must be satisfied:

$$\Delta \left[\pi_h F_H(s^*(p_h)) \left(U(p_h, H) - K \right) + (1 - \pi_h) F_L(s^*(p_h)) \left(U(p_h, L) - K \right) + \phi \right] \ge$$

$$\pi_h F_H(s^*(p_l)) \left(U(p_l, H) - K \right) + (1 - \pi_h) F_L(s^*(p_l)) \left(U(p_l, L) - K \right) + \phi \quad (C.21)$$

where $\Delta = 1$ if the IPO takes place also when B observes $\sigma = h$ and $\Delta = 0$ otherwise. Notice that the price p_h , in principle, need not be different from p_l if pooling or hybrid equilibria are possible. Consider first $\Delta = 1$. In this case both (C.19) and (C.20) must hold. This implies:

$$\pi_h F_H(s^*(p_h)) \left(U(p_h, H) - K \right) + (1 - \pi_h) F_L(s^*(p_h)) \left(U(p_h, L) - K \right) =$$

$$\pi_l F_H(s^*(p_l)) \left(U(p_l, H) - K \right) + (1 - \pi_l) F_L(s^*(p_l)) \left(U(p_l, L) - K \right)$$
(C.22)

Putting together (C.21) and (C.22), we obtain:

$$F_H(s^*(p_l))U(p_l, H) - F_L(s^*(p_l))U(p_l, L) \le -K\left[F_L(s^*(p_l)) - F_H(s^*(p_l))\right] \tag{C.23}$$

Note that, since trade occurs between B and I, lemma C.1 requires $p_l < u_H$. Lemmata C.2 and 2 then ensure that $p_l \ge v_H^{\phi} > u_L$. Given $u_L < p_l < u_H$, the LHS of (C.23) is strictly positive for all $s^* \in (\underline{s}, \overline{s})$. However, since $f_q(.)$ satisfies the monotone likelihood property, $[F_L(s^*) - F_H(s^*)] > 0$ for all $s^* \in (\underline{s}, \overline{s})$. This implies that the RHS of (C.23) is strictly negative. Hence, (C.23) is never satisfied. There is no equilibrium in which the IPO takes place for both $\sigma = l$ and $\sigma = h$.

Assume now $\Delta = 0$. Since trade occurs when $\sigma = l$, the interim participation constraint (C.20) must be satisfied. One can then verify that (C.21) and (C.20) imply that (C.23) should hold also in this case, so that the same argument used for $\Delta = 1$ applies.

To summarize, given that trade never occurs when B observes $\sigma = l$, any equilibrium with trade must be separating: when $\sigma = h$, B goes ahead with the IPO, and offers a price p_h at which trade occurs with positive probability. When $\sigma = l$, B does not go ahead with the IPO. (Equivalently, B goes ahead but offers a price $p_l \leq v_L$, i.e. a price that is never accepted by S). We now show that this is indeed the case by verifying that, when $\sigma = l$, B has no incentive to mimic and set p_h . Forgoing the IPO is incentive compatible if:

$$\pi_l F_H(s^*(p_h)) \left(U(p_h, H) - K \right) + (1 - \pi_l) F_L(s^*(p_h)) \left(U(p_h, L) - K \right) + \phi < 0$$
 (C.24)

Substituting ϕ from C.19 and rearranging yields:

$$F_H(s^*(p_h))U(p_l, H) - F_L(s^*(p_h))U(p_l, L) \ge -K\left[F_L(s^*(p_h)) - F_H(s^*(p_h))\right] \tag{C.25}$$

Applying a similar logic to that for (C.23), this is always satisfied for $u_L < p_h < u_H$. We now show that p_h must be in this range. Since trade occurs between B and I, lemma C.1 ensures that $p_h < u_H$. Lemma C.2 ensures $p_h \ge v_H^{\phi}$. Given lemma 2, this implies $p_h > u_L$. \square

The next lemma characterizes I's best reply given the separating equilibrium considered.

Lemma C.6. Given the equilibrium price p_h , I's equilibrium threshold $s^*(.)$ solves

$$\pi_h f_H(s^*(p_h)) U(p_h, H) + (1 - \pi_h) f_L(s^*(p_h)) U(p_h, L) = 0$$
 (C.26)

Proof. This follows from lemma C.3, given $u_L < p_h < u_H$ and $p_h \ge v_H^{\phi}$. The threshold $s^*(p_h)$ solves

$$\lambda f_H(s^*(p_h)) \Pr(p_h|H) U(p_h, H) + (1 - \lambda) f_L(s^*(p_h)) \Pr(p_h|L) U(p_h, L) = 0$$
 (C.27)

In the separating equilibrium considered, $\Pr(p_h|H) = \eta$ and $\Pr(p_h|L) = 1 - \eta$. Dividing by $\eta \lambda + (1 - \lambda)(1 - \eta)$ and rearranging, one obtains (C.26). \square

From lemma 2, $v_H^{\phi} > u_L$. As a result, candidate equilibria are characterized by p_h belonging to the continuum $[v_H^{\phi}, u_H)$. We now show that only $p = v_H^{\phi}$ survives NWBR.

Lemma C.7. The unique offering price that survives NWBR is $p_h = v_H^{\phi}$.

Proof. To prove that the unique offering price passing NWBR is $p_h = v_H^{\phi}$, we show that any situation where $p_h > v_H^{\phi}$ would be dominated. If I has refined beliefs, B could be better off by decreasing p_h .

To see this, suppose that the equilibrium is such that $p_h > v_H^{\phi}$. Recall that, since $\phi > 0$, $v_H^{\phi} > u_L$. Consider then a deviation \tilde{p} such that $v_H^{\phi} < \tilde{p} < p_h$. I replies by using threshold \tilde{s} . When observing h, B benefits from the deviation if:

$$\pi_{h}F_{H}(\tilde{s})U(\tilde{p},H) + (1-\pi_{h})F_{L}(\tilde{s})U(\tilde{p},L) - K\left[\pi_{h}F_{H}(\tilde{s}) + (1-\pi_{h})F_{L}(\tilde{s})\right] > \pi_{h}F_{H}(s^{*})U(p_{h},H) + (1-\pi_{h})F_{L}(s^{*})U(p_{h},L) - K\left[\pi_{h}F_{H}(s^{*}) + (1-\pi_{h})F_{L}(s^{*})\right]$$
 (C.28)

When observing l, B (weakly) benefits if:

$$\pi_l F_H(\tilde{s}) U(\tilde{p}, H) + (1 - \pi_l) F_L(\tilde{s}) U(\tilde{p}, L) - K \left[\pi_l F_H(s^*) + (1 - \pi_l) F_L(s^*) \right] + \phi \ge 0$$
 (C.29)

Substituting ϕ from condition (C.19) – the interim participation constraint for B when $\sigma = h$ – one obtains:

$$\pi_{l}F_{H}(\tilde{s})U(\tilde{p},H) + (1-\pi_{l})F_{L}(\tilde{s})U(\tilde{p},L) - K\left[\pi_{l}F_{H}(\tilde{s}) + (1-\pi_{l})F_{L}(\tilde{s})\right] \geq \\ \geq \pi_{h}F_{H}(s^{*})U(p_{h},H) + (1-\pi_{h})F_{L}(s^{*})U(p_{h},L) - \\ - K\left[\pi_{h}F_{H}(s^{*}) + (1-\pi_{h})F_{L}(s^{*})\right]$$
 (C.30)

Given $\tilde{p} > v_H^{\phi} > u_L$, then $U(\tilde{p}, L) < 0$. From $\tilde{p} < p_h < u_H$, it follows that $U(\tilde{p}, H) > 0$. Since $\pi_h > \pi_l$, the LHS of (C.28) is greater than the LHS of (C.30). Hence, if B weakly benefits from the deviation upon observing l, then she *strictly* benefits from the deviation upon observing h. Following a deviation to a lower price, the investor should then infer that it comes from B having received signal h. Upon observing such a deviation, I's threshold $\tilde{s}(\tilde{p})$ is therefore equal to $s^*(\tilde{p})$.

We now show that, given that I's threshold function stays the same for all $\tilde{p} \in [v_H^{\phi}, p_h]$, B has an incentive to deviate to a lower price whenever the participation constraint of the

type H issuer is not binding. To see this, note that differentiating B's payoff with respect to \tilde{p} yields:

$$\{\pi_{h}f_{H}(s^{*})\left(U(\tilde{p},H)-K\right)+(1-\pi_{h})f_{L}(s^{*})\left(U(\tilde{p},L)-K\right)\}\frac{ds^{*}(\tilde{p})}{d\tilde{p}}+ + \pi_{h}F_{H}(s^{*})\frac{dU(\tilde{p},H)}{d\tilde{p}}+(1-\pi_{h})F_{L}(s^{*})\frac{dU(\tilde{p},L)}{d\tilde{p}}$$
(C.31)

The last two terms are strictly negative. What about the first term?

From lemma C.6, s^* solves (C.26). Hence, the first term in (C.31) can be rewritten as:

$$-K (\pi_h f_H(s^*) + (1 - \pi_h) f_L(s^*)) \frac{ds^*(\tilde{p})}{d\tilde{p}}$$
 (C.32)

which is negative whenever $s^*(\tilde{p})$ is increasing in \tilde{p} . Rearranging (C.26), we see that s^* solves

$$\frac{f_H(s^*)}{f_L(s^*)} = -\frac{1 - \pi_h}{\pi_h} \frac{U(\tilde{p}, L)}{U(\tilde{p}, H)}$$
(C.33)

so that

$$\frac{ds^*(\tilde{p})}{d\tilde{p}} = -\frac{1 - \pi_h}{\pi_h} \frac{d\left(\frac{U(\tilde{p}, L)}{U(\tilde{p}, H)}\right) / d\tilde{p}}{d\left(\frac{f_H(s^*)}{f_L(s^*)}\right) / ds^*} > 0$$
 (C.34)

Hence, B's expected payoff is decreasing in the offering price.

This proves that, in a NWBR-refined equilibrium with trade, the price p_h must be equal to the minimum price that satisfies the participation constraint of the high quality issuer: $p_h = v_H^{\phi}$. \square

C.4 Proof of lemma 4

The set Φ is the set of values of ϕ such that: i) trade between B and I occurs with positive probability; ii) B makes zero profits in expectation.

Let Φ_T denote the set of values for ϕ such that trade between B and I occurs with positive probability. Given that in equilibrium $p_h = v_H^{\phi}$, any $\phi \in \Phi_T$ must satisfy $s^*(v_H^{\phi}) < \overline{s}$. This occurs if and only if $v_H^{\phi} < u_H$. Hence,

$$\Phi_T \equiv \{\phi : v_H^{\phi} < u_H\} \tag{C.35}$$

Let

$$\Phi_Z \equiv \{\phi : \pi_h F_H(s^*(v_H^{\phi})) U(v_H^{\phi}, H) + (1 - \pi_h) F_L(s^*(v_H^{\phi})) U(v_H^{\phi}, L) - K \left[\pi_h F_H(s^*(p_l)) + (1 - \pi_h) F_L(s^*(p_l))\right] + \phi = 0\}$$
(C.36)

denote the set of values of ϕ such that B makes zero profits. Clearly, $\Phi = \Phi_T \cap \Phi_Z$. We start by determining conditions under which the intersection of Φ_T and Φ_Z is non-empty. Then

we turn to uniqueness. Φ is non-empty if there exists $\phi \in \Phi_Z$ such that $v_H^{\phi} < u_H$. Using the identity $\phi = V(v_H^{\phi}, H)$, the equation in definition (C.36) can be rewritten as:

$$\pi_h F_H(s^*(v_H^{\phi})) U(v_H^{\phi}, H) + (1 - \pi_h) F_L(s^*(v_H^{\phi})) U(v_H^{\phi}, L) - K[\pi_h F_H(s^*(p_l)) + (1 - \pi_h) F_L(s^*(p_l))] + V(v_H^{\phi}, H) = 0$$
(C.37)

Since v_H^{ϕ} is an increasing function of ϕ , finding values of v_H^{ϕ} for which (C.37) is satisfied is equivalent to finding values of ϕ for which it is satisfied.

The upper limit for v_H^{ϕ} is u_H . For $v_H^{\phi} \leq v_H$, the LHS of (C.37) is negative. This is because: (1) the first line of (C.37) is negative (this can be shown by using the optimal condition for I's threshold – lemma C.6), (2) $K \geq 0$, and (3) for all $\phi \leq 0$ (equivalently, $v_H^{\phi} \leq v_H$), $V(v_H^{\phi}, H) \leq 0$. By continuity, therefore, if the LHS of (C.37) is positive when $v_H^{\phi} \to u_H$, then there exists a value $\phi \in \Phi_Z$ such that $v_H^{\phi} < u_H$. Consider then $v_H^{\phi} \to u_H$. The LHS of (C.37) converges to:

$$(1 - \pi_h)U(u_H, L) + V(u_H, H) - K \tag{C.38}$$

This proves the first statement of lemma 4.

We now prove the second part of the lemma. To do this is sufficient to show that the LHS of (C.37) is increasing in v_H^{ϕ} . Differentiating the LHS of (C.37):

$$\pi_{h}[f_{H}(s^{*}(v_{H}^{\phi}))U(v_{H}^{\phi},H) + (1-\pi_{h})f_{L}(s^{*}(v_{H}^{\phi}))U(v_{H}^{\phi},L)]\frac{ds^{*}(v_{H}^{\phi})}{dv_{H}^{\phi}} + \\
\pi_{h}F_{H}(s^{*}(v_{H}^{\phi}))\frac{dU(v_{H}^{\phi},H)}{dv_{H}^{\phi}} + (1-\pi_{h})F_{L}(s^{*}(v_{H}^{\phi}))\frac{dU(v_{H}^{\phi},L)}{dv_{H}^{\phi}} + \\
+\frac{dV(v_{H}^{\phi},H)}{dv_{H}^{\phi}} - K\left[\pi_{h}f_{H}(s^{*}(v_{H}^{\phi})) + (1-\pi_{h})f_{L}(s^{*}(v_{H}^{\phi}))\right]\frac{ds^{*}(v_{H}^{\phi})}{dv_{H}^{\phi}}$$
(C.39)

From the characterization of the optimal threshold for I in lemma C.6, $s^*(v_H^{\phi})$ is such that the first term is zero. Given A4, V(p, H) + U(p, H) is independent of p and therefore:

$$\frac{dV(v_H^{\phi}, H)}{dv_H^{\phi}} = -\frac{dU(v_H^{\phi}, H)}{dv_H^{\phi}} \tag{C.40}$$

It follows that:

$$\pi_{h} \frac{dU(v_{H}^{\phi}, H)}{dv_{H}^{\phi}} + (1 - \pi_{h}) \frac{dU(v_{H}^{\phi}, L)}{dv_{H}^{\phi}} + \frac{dV(v_{H}^{\phi}, H)}{dv_{H}^{\phi}} =$$

$$= (1 - \pi_{h}) \left[\frac{dU(v_{H}^{\phi}, L)}{dv_{H}^{\phi}} - \frac{dU(v_{H}^{\phi}, H)}{dv_{H}^{\phi}} \right]$$
(C.41)

Given that the RHS of (C.41) is non-negative by assumption, the LHS must also be non-negative. Since U(p,q) is decreasing in p and $F_q(s^*(v_H^{\phi})) < 1$ for q = H, L, the sum of

the second, third, and fourth term in (C.39) is positive. The last term is is negative, but becomes small as $K \to 0$. By continuity, for K sufficiently small, B's expected payoff is strictly increasing in v_H^{ϕ} . Hence, for K sufficiently small, we know that if a ϕ exists that satisfies (C.37), then it is unique. \square

C.5 Proof of Proposition 3

Consider the equilibrium described in lemma 3. We want to show that, evaluated from I's perspective, shares in the hands of I are underpriced and shares in the hands of B are overpriced. The expected net gain from the shares conditional on I choosing to buy them is:

$$\frac{\pi_h[1 - F_H(s^*(v_H^{\phi})]U(v_H^{\phi}, H) + (1 - \pi_h)[1 - F_L(s^*(v_H^{\phi})]U(v_H^{\phi}, L)}{\pi_h[1 - F_H(s^*(v_H^{\phi})] + (1 - \pi_h)[1 - F_L(s^*(v_H^{\phi})]}$$
(C.42)

From lemma C.6, I follows a threshold strategy that depends on v_H^{ϕ} . The threshold $s^*(v_H^{\phi})$ must be such that:

$$\pi_h f_H(s) U(p_h, H) + (1 - \pi_h) f_L(s) U(p_h, L) > 0$$
 (C.43)

for all $s > s^*(v_H^{\phi})$. Integrating (C.43) between $s^*(v_H^{\phi})$ and \overline{s} shows that shares bought by I are on average underpriced.

We now turn attention to the case in which I does not buy and B holds the shares. The expected net gain from the shares conditional on I choosing not to buy them is:

$$\frac{\pi_h F_H(s^*) U(p_h, H) + (1 - \pi_h) F_L(s^*) U(p_h, L)}{\pi_h F_H(s^*) + (1 - \pi_h) F_L(s^*)}$$
(C.44)

The threshold $s^*(v_H^{\phi})$ must be such that:

$$\pi_h f_H(s) U(p_h, H) + (1 - \pi_h) f_L(s) U(p_h, L) < 0$$
 (C.45)

for all $s < s^*(v_H^{\phi})$. Integrating (C.45) between \underline{s} and $s^*(v_H^{\phi})$ shows that (C.44) is negative. \Box

C.6 Proof of Proposition 4

Consider the equilibrium discussed in lemma 3. Total expected net gains from the shares (whether bought by I or not) are:

$$\pi_h U(v_H^{\phi}, H) + (1 - \pi_h) U(v_H^{\phi}, L)$$
 (C.46)

We first show that (C.46) is positive when the precision of B's information is extremely high and then we turn to the case in which I's signal s is almost independent of q. Given the equilibrium in lemma 3, trade between I and B occurs if and only if the price v_H^{ϕ} does not exceed I's reservation price for a type H issuer: $v_H^{\phi} < u_H$. This is necessary and sufficient

for $U(v_H^{\phi}, H) > 0$. It is then clear that, for π_h close enough to unity, (C.46) is positive if and only if $v_H^{\phi} < u_H$. This establishes the first claim in the proposition. We now turn to the second.

From lemma C.6, I follows a threshold strategy that depends on v_H^{ϕ} . The threshold $s^*(v_H^{\phi})$ must be such that:

$$\pi_h f_H(s) U(v_H^{\phi}, H) + (1 - \pi_h) f_L(s) U(v_H^{\phi}, L) > 0$$
 (C.47)

for all $s > s^*(v_H^{\phi})$. Integrating (C.47) between $s^*(v_H^{\phi})$ and \overline{s} shows that:

$$\pi_h[1 - F_H(s^*(v_H^{\phi})]U(v_H^{\phi}, H) + (1 - \pi_h)[1 - F_L(s^*(v_H^{\phi})]U(v_H^{\phi}, L) > 0$$
 (C.48)

If $s \to^d \tilde{s}$, then $F_H(.)$ and $F_L(.)$ converge to the same distribution $\tilde{F}(.)$. For trade to occur with positive probability, \tilde{F} evaluated at the threshold must be less than one. But then the LHS of expression (C.48) reduces to (C.46) whenever the probability of trade is positive implying that, if trade occurs with positive probability, shares are underpriced. We now show that if shares are underpriced then trade occurs. Expression (C.47) evaluated at $s = s^*(v_H^\phi)$ is equal to zero. Solving it for the likelihood ratio $f_H(s^*)/f_L(s^*)$ shows that $f_H(s^*)/f_L(s^*) < 1$ whenever (C.46) is positive. The threshold s^* is thus bounded above by some \hat{s} defined as the value of s such that $f_H(\hat{s})/f_L(\hat{s}) = 1$. Since $\hat{s} < \overline{s}$, it follows that $s^* < \overline{s}$. Hence, given underpricing, trade occurs with positive probability. \Box

D Proofs of Claims Made in Section 8 (Material not meant for publication)

Information Structure

Claim 1. When B is perfectly informed, there is a continuum of zero-profit equilibria where: (i) the IPO takes place only when q = H, (ii) $\phi = 0$, (iii) the offering price is in the interval $[v_H, u_H]$, and (iv) I buys with probability one.

Proof. Suppose that $\phi = 0$. When q = L, B has no incentive to undertake the IPO: if she sells the shares with probability one, she earns zero profits, while if she doesn't sell the shares with probability one, she makes losses. Therefore, when q = L, not undertaking the IPO is a best reply for B. Now consider q = H. Let the equilibrium price selected when q = H be $p^* \in [v_H, u_H]$. It is clear that, given p^* , purchasing the shares with probability one is optimal for I. Since at $p = p^*$ B sells the shares with probability one, she makes neither losses nor gains from underwriting. Hence, $\phi = 0$ guarantees zero profits. What about B's pricing incentives? Suppose that, when he observes an out of equilibrium price $p \in [v_H, u_H]$, I purchases the shares with probability one. Then it is clear that setting $p = p^*$ when q = H is optimal for B. Setting $p > u_H$ would result in B keeping the high-quality shares for sure, but would also entail losses. Setting $p < v_H$ would not satisfy the H-type issuer's participation constraint. Finally, we need to verify that the proposed out of equilibrium strategy for I – namely, that when he observes an out of equilibrium price $p \in [v_H, u_H]$,

I purchases the shares with probability one – does not violate NWBR. To see that this is indeed the case, consider an out of equilibrium price $p \in [v_H, u_H]$. Suppose that B selects p and that, upon observing p, I uses a threshold s^D . B's payoff from deviating to p having observed q = L is:

$$(1 - F_L(s^D))(U(p, L) - K) \le 0 (D.1)$$

Hence, for q = L, B can only (weakly) lose from deviating to p. Two cases may then arise: (1) B loses from deviating to p both when q = L and when q = H, or (2) B only loses from deviating to p when q = L. In both cases, beliefs such that the deviation emanates from B having observed q = H are not ruled out by NWBR. \square

Claim 2. When B is entirely uninformed, her expected payoff is decreasing in p.

Proof. Upon selecting a price $p \in [v_H, u_H]$, B's expected payoff is

$$\lambda F_H(s^*(p)) (U(p, H) - K) + (1 - \lambda) F_L(s^*(p)) (U(p, L) - K) + \phi$$
 (D.2)

The derivative of (D.2) with respect to p is

$$[\lambda f_{H}(s^{*}(p))U(p,H) + (1-\lambda)f_{L}(s^{*}(p))U(p,L)] \frac{ds^{*}(p)}{dp} -K \frac{ds^{*}(p)}{dp} [\lambda f_{H}(s^{*}(p)) + (1-\lambda)f_{L}(s^{*}(p))]$$

$$+ \left[\lambda F_{H}(s^{*}(p)) \frac{dU(p,H)}{dp} + (1-\lambda)F_{L}(s^{*}(p)) \frac{dU(p,L)}{dp}\right]$$
(D.3)

From the definition of $s^*(p)$, the first term in (D.3) is equal to zero. Since $\frac{ds^*(p)}{dp} > 0$, the second term in (D.3) is negative. Finally, from A3(i), $\frac{dU(p,q)}{dp} < 0$ for both q = H, L. Hence, the third expression in (D.3) is negative, which proves our claim. \square

Equilibria that fail NWBR

Claim 3. When B is perfectly informed, then a situation where the IPO takes place only when the issuer is of type H (in which case the investor buys with probability one) is an equilibrium.

Proof. This trivially follows from claim 1. \square

Claim 4. When the intermediary is imperfectly informed, perfect Bayesian equilibria with zero profits involve full revelation of the intermediary's information through the price choice (the intermediary goes on with IPO only when she receives favorable information).

Proof. Consider the proof of lemma 3. Lemma C.5 implies the result. Notice that lemma C.5 does not require NWBR. Hence, in all perfect Bayesian equilibria with zero profits, the intermediary goes on with the IPO only when she observes $\sigma = h$. \square

Low quality should be traded

Claim 5. Existence of "efficient equilibria" when the intermediary is perfectly informed.

Proof. Consider a situation in which a type q issuer sells whenever $p \geq v_q$, the intermediary announces some $p_q \in [v_q, u_q]$, and the investor buys with probability one at p_q . Clearly enough, S and I are playing best replies. Assume that I's beliefs assign probability one to type H for all out of equilibrium prices in the interval $[v_H, u_H]$ and probability one to type L for all prices lower than v_H . These ensure that setting p_q is a best reply for the intermediary. When $u_L \geq v_H$, the intermediary would sell with probability one at all prices in the interval $[v_L, u_H]$. When $v_H > u_L$ she would sell with probability one at all prices in the intervals $[v_L, u_L]$ and $[v_H, u_H]$. At these prices, her payoff would be equal to ϕ independently of the price she announces. When $v_H > u_L$, B would sell with probability zero at all prices in the interval (u_L, v_H) . Given S's strategy, q = L at all prices (u_L, v_H) so that B has no incentive to deviate to these prices. Hence, given I's beliefs, announcing $p_q \in [v_q, u_q]$ is a best reply for B. Zero profits then requires $\phi = 0$. We now show that I's beliefs are compatible with NWBR. B's payoff from deviating to any $p \in [v_H, u_H]$ having observed q is:

$$(1 - F_q(s^D))[U(p,q) - K]$$
 (D.4)

where s^D is I's threshold upon observing p. Since U(p,H) > U(p,L), if (D.4) is weakly positive for q=L, then it is strictly positive for q=H. Hence, beliefs such that the deviation emanates from B having observed q=H are compatible with NWBR. Finally, if all qualities generate gains from trade, welfare is maximized when the amount of trade is maximized. Hence, these equilibria are efficient. \square

References

- [1] Aggarwal, R., Krigman, L., and Womack, K. 2002. Strategic IPO underpricing, Information Momentum, and Lockup Expiration Selling. Journal of Financial Economics. 66: 105-137.
- [2] Akerlof, G.A., 1970. The Market for "Lemons": Quality Uncertainty and the Market Mechanism. Quarterly Journal of Economics. 84: 488-500.
- [3] Allen, F., and Faulhaber, G., 1989. Signaling by Underpricing in the IPO Market. Journal of Financial Economics. 23: 303-323.
- [4] Baron, D.P., 1982. A Model of the Demand for Investment Banking Advising and Distribution Services for New Issues. Journal of Finance. 37: 955-976.
- [5] Baron, D.P. and Hölmstrom, B., 1980. The Investment Banking Contract for New Issues Under Asymmetric Information: Delegation and the Incentive Problem. Journal of Finance. 35: 1115-1138.
- [6] Bénabou, R., and Tirole, J., 2003. Intrinsic and Extrinsic Motivation. Review of Economic Studies. 70: 489-520.
- [7] Benveniste, L.M., and Spindt, P.A., 1989. How Investment Bankers Determine the Offer Price and Allocation of New Issues. Journal of Financial Economics. 24: 343-361.
- [8] Biais, B., Bossaerts, P., and Rochet, J.-C., 2002. An Optimal IPO Mechanism. Review of Economic Studies. 69: 117-146.
- [9] Brau J.C., Francis, B., and Kohers, N., 2003. The choice of IPO versus takeover: empirical evidence. Journal of Business 76: 583-612.
- [10] Brau J.C. and Fawcett, S.E., 2006 Initial Public Offerings: an analysis of theory and practice. Journal of Finance 61: 399-436.
- [11] Chemmanur, T.J. 1993. The Pricing of Initial Public Offerings: a Dynamic Model with Information Production. Journal of Finance. 48: 285-304.
- [12] Chemmanur, T.J., and Fulghieri, P., 1994. Investment Bank Reputation, Information Production, and Financial Intermediation. Journal of Finance. 49: 57-79.
- [13] Chemmanur, T.J. and Fulghieri, P., 1999. A theory of the going-public decision. Review of Financial Studies. 12: 249-279.
- [14] Chen, H-C., and Ritter, J.R., 2000. The Seven Percent Solution. Journal of Finance. 55: 1105-1131.
- [15] Cho, I. K., and Kreps, D.M., 1987. Signaling Games and Stable Equilibria. Quarterly Journal of Economics. 102: 179-221.

- [16] Cornelli, F., and Goldreich, D., 2003. Bookbuilding: How Informative Is the Order Book? Journal of Finance. 58: 1415-44.
- [17] Demers, E. A., and Lewellen, K., 2003. The Marketing Role of IPOs: Evidence from Internet Stocks. Journal of Financial Economics. 68: 413-437.
- [18] Ellingsen, T., 1997. Price Signals Quality: The Case of Perfectly Inelastic Demand. International Journal of Industrial Organization. 16: 43-61.
- [19] Ellis, K., Michaely, R., and O'Hara, M., 1999. A guide to the initial public offering process. Corporate Finance Review 3: 14-18.
- [20] Faure-Grimaud, A., and Gromb, D., 2004. Public Trading and Private Incentives. Review of Financial Studies. 17: 985-1014.
- [21] Fudenberg, D. and Tirole, J., 1991. Game Theory. MIT Press, Cambridge MA.
- [22] Glosten, L.R., and Milgrom, P.R., 1985. Bid, ask and transaction prices in a specialist market with heterogeneously informed traders. Journal of Financial Economics. 14: 71-100.
- [23] Grinblatt, M., and Hwang, C.-Y., 1989. Signaling and the pricing of new issues. Journal of Finance. 44: 393-420.
- [24] Habib, M., and Ljungqvist, A., 2001. Underpricing and Entrepreneurial Losses in IPOs: Theory and Evidence. Review of Financial Studies. 14: 433-458.
- [25] Hansen, R., 2001. Do investment banks compete in IPOs? The advent of the 7% plus contract. Journal of Financial Economics. 59: 313-46.
- [26] Hughes, P., and Thakor, A., 1992. Litigation Risk, Intermediation, and the Underpricing of Initial Public Offerings. Review of Financial Studies. 5: 709-742.
- [27] Loughran, T., and Ritter, J.R., 2002. Why don't Issuers Get Upset About Leaving Money on the Table in IPOs? Review of Financial Studies. 15: 413-443.
- [28] Mello, A.S., and Parsons, J.E., 2000. Hedging and liquidity. Review of Financial Studies 13,127-153.
- [29] Michaely, R., and Shaw, W.H., 1994. The Pricing of Initial Public Offering: Tests of Adverse-Selection and Signaling Theories. Review of Financial Studies. 7: 279-319.
- [30] Muscarella, C.J., and Vetsuypens, M. R., 1989. A simple test of Baron's model of IPO underpricing. Journal of Financial Economics. 24: 125-135.
- [31] Pagano, M., Panetta, F., and Zingales, L., 1998. Why do Firms Go Public? An Empirical Analysis. Journal of Finance. 53: 27-64.
- [32] Ritter, J. R., 2003. Investment Banking and Securities Issuance. (In) Handbook of the Economics of Finance (eds. Constantinides, Harris, and Stulz).

- [33] Ritter, J.R., and Welch, I., 2002. A Review of IPO Activity, Pricing, and Allocations. Journal of Finance. 57: 1795-1828.
- [34] Rock, K., 1986. Why new issues are underprized. Journal of Financial Economics. 15: 187-212.
- [35] Sherman, A. E., 1999. Underwriter Certification and the Effect of Shelf Registration on Due Diligence. Financial Management. 28: 5-19.
- [36] Tirole, J., 2006 The Theory of Corporate Finance. Princeton University Press. Princeton and Oxford.
- [37] Welch, I., 1989. Seasoned Offerings, Imitation Costs, and the Underpricing of Initial Public Offerings. Journal of Finance. 44: 421-449.
- [38] Zingales, L., 1995. Insider Ownership and the Decision to Go Public. Review of Economic Studies. 62: 425-448.

Ultimi Contributi di Ricerca CRENoS

I Paper sono disponibili in: http://www.crenos.it

- 07/13 Giovanni Sulis, "What Can Monopsony Explain Of The Gender Wage Differential In Italy?"
- **07/12** Gerardo Marletto, "Crossing The Alps: Three Transport Policy Options"
- 07/11 Sergio Lodde "Human Capital And Productivity Growth In The Italian Regional Economies: a Sectoral Analysis"
- 07/10 Axel Gautier, Dimitri Paolini, "Delegation, Externalities And Organizational Design"
- 07/09 Rinaldo Brau, Antonello E. Scorcu, Laura Vici, "Assessing visitor satisfaction with tourism rejuvenation policies: the case of Rimini, Italy"
- 07/08 Dimitri Paolini, "Search and the firm's choice of the optimal labor contract"
- 07/07 Giacomo Carboni, "Shape of U.S. business cycle and long-run effects of recessions"
- 07/06 Gregory Colcos, Massimo Del Gatto, Giordano Mion and Gianmarco I.P. Ottaviano, "Productivity and firm selection: intra-vs international trade"
- 07/05 Silvia Balia, "Reporting expected longevity and smoking: evidence from the share"
- 07/04 Raffaele Paci, Stefano Usai, "Knowledge flows across European regions"
- **07/03** Massimo Del Gatto, Giordano Mion and Gianmarco I.P. Ottaviano, "Trade Integration, firm selection and the costs of non-europe"
- **07/02** *Vittorio Pelligra*, "Intentions, trust and frames: a note on sociality and the theory of games"
- **07/01** Oliviero A. Carboni and Giuseppe Medda, "Government size and the composition of public spending in a neoclassical growth model"
- **06/17** Susana Iranzo, Fabiano Schivardi and Elisa Tosetti, "Skill dispersion and firm productivity: an analysis with employer-employee matched data"
- **06/16** Luigi Guiso and Fabiano Schivardi, "What determines entrepreneurial clusters?"
- **06/15** *Vittorio Pelligra*, "Trust Responsiveness: on the dynamics of fiduciary interactions"
- **06/14** Rinaldo Bran, Alessandro Lanza and Francesco Pigliaru, "How fast are small tourist countries growing? The 1980-2003 evidence"
- 06/13 Emanuela Marrocu and Raffaele Paci, "The effects of public capital on the productivity of the italian regions"
- **06/12** Raffaele Paci and Stefano Usai, "Agglomeration economies and growth-the case of Italian local labour systems, 1991-2001"
- 06/11 Fabrizio Ariani and Luca G. Deidda, "The Monopolist's blues"
- **06/10** Bianca Biagi, Dionysia Lambiri and Vicente Roynela, "Quality of life in the economic and urban economic literature"
- **06/09** Manuela Pulina and Bianca Biagi, "Tourism, environmental quality and economic growth: empirical evidence and policy implications"

Finito di stampare nel mese di Dicembre 2007 Presso Editoria&Stampa Zona Industriale Predda Niedda str. n. 10 07100 Sassari

